

Recommendation for Case Closure

Former Atlantic Richfield Company

Station No. 11102

100 MacArthur Boulevard Oakland, California 94610

ACEH Case # RO0000456

RECEIVED

4:56 pm, Mar 13, 2012 Alameda County

Environmental Health

"I declare that to the best of my knowledge at the present time, that the information and/or recommendations contained in the attached document are true and correct."

Submitted by:

ARCADIS U.S., Inc

HETRillips

Hollis E. Phillips, P.G. Project Manager/Principal Geologist



ARCADIS U.S., Inc. 100 Montgomery Street, Suite 300 San Francisco, CA 94104 Tel 415.374.2744 Fax 415.374.2745 www.arcadis-us.com

ENVIRONMENT

Date: March 12, 2012

Contact: Hollis E. Phillips

Phone: 415.374.2744 ext 13

Email: Hollis.Phillips@arcadisus.com

Our ref: GP09BPNA.C111



g:\projects\bp_arco\11102 oakland\closure\perjury statment.docx



Imagine the result

Atlantic Richfield Company

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102 100 MacArthur Boulevard Oakland, California 94610 ACEH Case # RO0000456

March 12, 2012

Steve Rice

Ben McKenna

HETRillips

Hollis E. Phillips, PG Project Manager/Principal Geologist



Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

Prepared for: British Petroleum

Prepared by: ARCADIS U.S., Inc. 2033 North Main Street Suite 340 Walnut Creek California 94596 Tel 925.274.1100 Fax 925.274.1103

Our Ref.: GP09BPNA.C111

Date: March 12, 2012

This document is intended only for the use of the individual or entity for which it was prepared and may contain information that is privileged, confidential and exempt from disclosure under applicable law. Any dissemination, distribution or copying of this document is strictly prohibited.

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

| 1. | . Introduction | | | | | | | | | | |
|----|----------------|-----------|--|----|--|--|--|--|--|--|--|
| | 1.1 | Site Ba | ckground | 2 | | | | | | | |
| | 1.2 | Site Ge | eology and Hydrogeology | 2 | | | | | | | |
| 2. | Previou | IS Site | Investigations | 3 | | | | | | | |
| | | 2.1.1 | UST Removal – 1988 | 3 | | | | | | | |
| | | 2.1.2 | Monitoring Well Installation – 1989 | 4 | | | | | | | |
| | | 2.1.3 | UST Removal/Replacement and Station Upgrades – 1990 | 4 | | | | | | | |
| | | 2.1.4 | Soil and Groundwater Investigation – 1994 | 4 | | | | | | | |
| | | 2.1.5 | Well Recovery Test – 1999 | 5 | | | | | | | |
| | | 2.1.6 | Additional Soil & Groundwater Investigation – 2005 | 5 | | | | | | | |
| | | 2.1.7 | Additional Monitoring Well Installation – 2010 | 7 | | | | | | | |
| 3. | Extent | 7 | | | | | | | | | |
| | 3.1 | Soil Co | Soil Conditions | | | | | | | | |
| | 3.2 | Ground | 8 | | | | | | | | |
| | | 3.2.1 | TPHg | 8 | | | | | | | |
| | | 3.2.2 | Benzene | 9 | | | | | | | |
| | | 3.2.3 | МТВЕ | 9 | | | | | | | |
| | 3.3 | Separa | te-Phase Hydrocarbon Status | 10 | | | | | | | |
| | 3.4 | Hydrau | lic Gradient Trends | 10 | | | | | | | |
| 4. | Benefic | ial Use | 25 | 10 | | | | | | | |
| | 4.1 | San Fra | ancisco Bay RWQCB Basin Plan | 10 | | | | | | | |
| | 4.2 | Sensitiv | ve Receptor Survey | 11 | | | | | | | |
| | 4.3 | Utility S | Survey | 11 | | | | | | | |
| | 4.4 | Enviror | 12 | | | | | | | | |
| | 4.5 | Summa | ary Factors Affecting Long-Term Fate of Contaminants | 13 | | | | | | | |
| 5. | Remed | ial Acti | vities | 15 | | | | | | | |
| 6. | Remed | ial Effe | ctiveness | 15 | | | | | | | |

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

| 7. | Conclusions | 15 |
|----|-------------|----|
| 8. | References | 17 |

Tables

| Table 1 | Historical Soil Results |
|---------|--|
| Table 2 | Historical Groundwater Results |
| Table 3 | Historical Groundwater Flow Direction and Gradients |
| Table 4 | Most Recent Maximum Concentration of Contaminants Detected in Soil and Groundwater |

Figures

| Figure 1 | Site Location Map |
|-----------|--|
| Figure 2 | Site Map with Soil Boring and Monitoring Well Locations |
| Figure 3 | Historical Lateral Extent of TPHg Soil Impacts |
| Figure 4 | Historical Lateral Extent of MTBE Soil Impacts |
| Figure 5 | Site Plan with Cross Section Locations |
| Figure 6 | Vertical Extent of TPHg, Benzene, and MTBE Soil Impacts; (Geologic Cross Section A-A') |
| Figure 7 | Vertical Extent of TPHg, Benzene, and MTBE Soil Impacts; (Geologic Cross Section B-B') |
| Figure 8 | Extent of TPHg Groundwater Impacts – February 2012 |
| Figure 9 | Extent of MTBE Groundwater Impacts – February 2012 |
| Figure 10 | Extent of TPHg Groundwater Impacts – August 2011 |
| Figure 11 | Extent of MTBE Groundwater Impacts – August 2011 |
| Figure 12 | Extent of TPHg Groundwater Impacts – February 2011 |
| Figure 13 | Extent of MTBE Groundwater Impacts – February 2011 |
| Figure 14 | Historical Groundwater Flow Direction Rose Diagram |
| Figure 15 | Potentiometric Surface Contours - August 2011 |
| Figure 16 | Potential Exposure Pathway Flow Chart |

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

Appendices

| Appendix A | Monitoring Well Construction Details and Soil Boring Logs |
|------------|--|
| Appendix B | Concentration versus Time Graphs for TPHg , Benzene and MTBE |
| Appendix C | Historical Underground Utility Figures |
| Appendix D | Alameda County Environmental Health Case Closure Summary |

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

Acronyms/Abbreviations

| ACEH | Alameda County Environmental Health |
|------------|---|
| Alton | Alton Geoscience Inc. |
| Alisto | Alisto Engineering Group |
| ARCADIS | ARCADIS U.S. |
| ARCO | Atlantic Richfield Company |
| BAI | Broadbent Associates, Inc |
| Basin Plan | San Francisco Bay Basin (Region 2) Water Quality Control Plan |
| bgs | below ground surface |
| BP | British Petroleum |
| BTEX | benzene, toluene, ethylbenzene and xylene |
| btoc | below top of casing |
| Cambria | Cambria Environmental Technology Inc. |
| сос | contaminant of concern |
| CPT | cone penetrometer testing |
| су | cubic yard |
| DTSC | Department of Toxic Substances Control |
| DTW | depth to water |
| DWR | Department of Water Resources |
| ft | feet |

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

| EBMUD | East Bay Municipal Utility District |
|---------|---|
| EMCON | EMCON Associates, Inc. |
| EPA | Environmental Protection Agency |
| ESL | Environmental Screening Level |
| Gph | gallons per hour |
| GSI | GeoStrategies, Inc |
| GWET | Groundwater extraction and treatment |
| HVOCs | Halogenated Volatile Organic Compounds |
| KEI | Kaprealian Engineering, Inc. |
| MDL | Method detection limit |
| mg/kg | milligram per kilogram |
| mg/L | milligram per liter |
| MTBE | methyl tertiary butyl ether |
| MW | monitoring well |
| Pacific | Pacific Environmental Group, Inc. |
| Ppb | parts per billion |
| Ppm | parts per million |
| Ppmv | parts per million volume |
| RESNA | RESNA Industries, Inc. |
| RWQCB | Regional Water Quality Control Board, San Francisco |

Bay

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

| SECOR | Secor International, Inc. |
|-------------|--|
| Site | Former Atlantic Richfield Company Station No. 11266, located at 1541 Park Street, Alameda, California |
| SPH | separate phase hydrocarbon |
| SRS | Sensitive receptor survey |
| ТВА | Tert-butyl alcohol |
| TCLP | Toxicity characteristic leaching procedure |
| TOG | total oil and grease |
| TPH | total petroleum hydrocarbons |
| TPHd | total petroleum hydrocarbons as diesel |
| TPHg | total petroleum hydrocarbons as gasoline |
| TPHo | total petroleum hydrocarbons as oil |
| URS | URS Corporation, Inc. |
| UST | underground storage tank |
| VOC | volatile organic compounds |
| Water Board | California Regional Water Quality Control Board |
| WDC | WDC Exploration & Wells |
| WET | Waste extraction test |
| µg/L | microgram per liter |

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

1. Introduction

ARCADIS has prepared this *Recommendation for Case Closure* (Report) for the Former Atlantic Richfield Company (ARCO) Service Station No. 11102 (Site) located at 100 MacArthur Boulevard in Oakland California (**Figure 1**). Case closure is warranted for the site based on the following information:

- Free product has never been observed in any of the monitoring wells or excavated areas on-site.
- Current groundwater concentrations are orders of magnitude below historical maximums, are continuing to decrease with time, or are not detected above the detection limit, indicating that natural attenuation processes are occurring at the Site.
- The Site has been adequately characterized.
- The low-level plume is decreasing in size.
- No sensitive receptors are likely to be impacted, including surface-water bodies, municipal wells and drinking water sources.
- The site presents no current or potential risk to human health or the environment.

The report is organized into the following sections:

- Section 1 provides the Site background and Geology/Hydrogeology.
- Section 2 provides the Previous Site Investigations.
- Section 3 presents the Extent of Soil and Groundwater Impacts.
- Section 4 provides the Beneficial Uses.
- Section 5 presents Remedial Activities.
- Section 6 presents Remedial Effectiveness.

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

- Section 7 presents Conclusions with Recommendations for Case Closure
- Section 8 presents References.

1.1 Site Background

The Site is located at 100 MacArthur Boulevard in Oakland, California. It is an active 76-branded gasoline station. BP acquired the property from Mobil Oil Corporation in 1989. Although BP sold the property to TOSCO Marketing Corporation in 1994, it retained the environmental liability for contamination released prior to this transfer. Current improvements to the Site include three fiberglass gasoline underground storage tanks (USTs) (6,000-gallons, 10,000-gallons, and 12,000-gallons) believed to have been installed in 1990, one 1,000-gallon double-walled fiberglass underground waste oil storage tank installed in 1988, two fuel dispenser islands with a total of eight dispensers, and a convenience store building with three vehicle service bays. The majority of the Site surface is paved with cement and asphalt. A Site Location Map is provided as **Figure 1**. A Site map is included as **Figure 2**.

The Site is bound by MacArthur Boulevard to the southwest, Oakland Avenue to the southeast, Harrison Street to the northwest and single-family residential dwellings to the northeast (uphill from the Site and its retaining wall). Interstate 580 and the associated on- and off-ramps are located across MacArthur Boulevard to the southwest. A small parking lot and several small commercial buildings are located across Oakland Avenue to the southeast. A Quik Stop retail gasoline station is located across Harrison Street to the northwest at 96 West MacArthur Boulevard. The Quik Stop gasoline station is Former Unocal Station No.1871, an active fuel leak case (ACEH Case No.RO0000455 / GeoTracker Global ID No.T0600101493).

1.2 Site Geology and Hydrogeology

The Site is situated at an approximate elevation of 90 feet above mean sea level. The Site is relatively flat, but slopes slightly to the west, consistent with the local topography. Sediments encountered at the Site consist primarily of silty clays or clayey silts with small lenses of sand and gravel, extending from the ground surface to the total depth investigated, approximately 36 ft bgs.

Throughout most of the Alameda County portion of the East Bay Plain, from Hayward north to Albany, water level contours show that the general direction of groundwater flow is from east to west or from the Hayward Fault to the San Francisco Bay.

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

Groundwater flow direction generally correlates to topography. Flow direction and velocity are also influenced by buried stream channels that typically are oriented in an east to west direction. Historic groundwater flow direction at the Site has varied between south and west/northwest, but has been predominantly southwest to west. The nearest natural drainage is Glen Echo Creek, located approximately 1,450 feet northwest of the Site. Glen Echo Creek flows generally northeast to southwest near the Site vicinity (BAI 2009).

2. Previous Site Investigations

Several investigations have been conducted at the site since 1988. Historical Soil Analytical Data is presented in **Table 1**. Historical Groundwater Analytical Data is presented in **Table 2**. Locations of soil borings and monitoring wells (MWs) are presented on **Figure 2**. The historical lateral extent of TPHg and benzene soil impacts in the vadose zone is provided on **Figures 3** and **4**, respectively. Geologic Cross Section Locations are presented on **Figure 5**. The vertical extent of TPHg and benzene soil impacts is provided on **Figure 6** and **7**. Monitoring Well Construction Details and Soil Boring Logs are presented in **Appendix A**

2.1.1 UST Removal - 1988

Kaprealian Engineering, Inc. (KEI) observed the removal of a steel underground waste oil storage tank, variously reported to have been of 550-gallon or 280-gallon capacity, on 19 September 1988. Work was performed for Mobil Oil Corporation prior to the sale of the property to BP. KEI reported that no holes or cracks were evident in the tank. However, upon UST removal a representative of Alameda County Environmental Health (ACEH) reportedly observed a hole in the UST and petroleum product "dripping" from the west sidewall (ERI 1998). Two soil samples were collected during the tank removal activities: sample WO was collected from the bottom of the tank pit and sample Comp WO was composed of two grab samples collected from the excavated soil stockpiled on-site, which equaled approximately 15 cubic yards (cy). Total petroleum hydrocarbons in the diesel range (TPHd) were reported at concentrations of 2.0 parts per million (ppm) and 1,700 ppm in samples WO and Comp WO, respectively. Total oil & grease (TOG) was reported at concentrations of 24 ppm and 65,000 ppm in samples WO and Comp WO, respectively. No volatile organic compounds (VOCs) were detected above the laboratory reporting limit in sample WO (KEI 1988).

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

2.1.2 Monitoring Well Installation - 1989

On 25 and 26 October 1989 Alton Geoscience, Inc. (Alton) observed the advancement of three soil borings onsite. Each boring was drilled to 33 feet below ground surface (ft bgs) and converted into 4-inch diameter groundwater monitoring wells (MW-1, MW-2, and MW-3). Refer to Figure 2 for MW locations. The wells were completed to a total depth of 32 ft bgs and screened from 12 ft bgs to total depth. Three soil samples were collected from each boring at depths of approximately 5 ft bgs, 10 ft bgs, and 15 ft bgs. Soil samples were analyzed for total petroleum hydrocarbons in the gasoline range (TPHg), and benzene, toluene, ethylbenzene, and total xylenes (collectively referred to as BTEX). Soil samples from boring MW-1 adjacent to the former waste oil UST were also analyzed for TPHd and TOG. Benzene was detected in soil samples collected from boring MW-2 at depths of 5 ft bgs and 10 ft bgs at 6 micrograms per kilogram (• g/kg) and 8 • g/kg, respectively. Toluene and total xylenes were detected in the 5 ft bgs soil sample collected from boring MW-3 at 6 • g/kg and 13 • g/kg, respectively. Wells MW-1, MW-2, and MW-3 were developed on 4 November 1989 and groundwater samples collected on 11 November 1989. Groundwater samples were analyzed for TPHg and BTEX, with samples from MW-1 also being analyzed for TOG and halogenated VOCs (HVOCs). The groundwater sample collected from well MW-1 contained benzene at 3.4 micrograms per liter (• g/L), toluene at 0.6 • g/L, and 1,2dichloroethane (1,2-DCA) at 0.9 • g/L. The groundwater sample collected from well MW-2 contained benzene at $6.5 \cdot g/L$. No other analytes were detected above their reporting limits (Alton 1989).

2.1.3 UST Removal/Replacement and Station Upgrades - 1990

The station underwent renovations in 1990 when new USTs, pump islands, and a new canopy were installed. Limited documentation was available regarding the renovation activities. No soil or groundwater investigation activities were conducted during the renovations.

2.1.4 Soil and Groundwater Investigation - 1994

On November 22, 1994 EMCON Associates, Inc. (EMCON) conducted a site assessment that consisted of collecting two discrete soil samples (TD-1 and TD-3) at approximately 0.5 feet bgs below the fuel dispensers at the northern and southern pump islands. Additionally, two soil borings, THP-1 and THP-2 were advanced to depth of 22 and 17.5 feet bgs, respectively with soil and groundwater samples collected. Refer to **Figure 2** for boring locations. .Soil samples were submitted for

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

analysis for TPHg, TPHd, total petroleum hydrocarbons as oil (TPHo), BTEX and HVOCs. TPHg was detected in the soil sample collected from THP-1 at 13 feet bgs at a concentration of 1.2 mg/Kg with no other concentrations reported from that boring. Soil analysis of THP-2 samples at depths of 7 and 10 feet bgs reported no concentrations above the laboratory detection limits. Soil sample analysis from TD-1 reported concentrations of TPHg at 1.4 mg/Kg, TPHd at 2,100 mg/Kg, toluene at 0.006 mg/Kg and total xylenes at 0.04 mg/Kg, no other concentrations were detected above the reporting limit in samples collected from TD-1. Soil sample analysis from TD-3 at 0.5 feet bgs reported concentrations of TPHd at 470 mg/Kg with no other analytes detected.

2.1.5 Well Recovery Test - 1999

Cambria Environmental Technology, Inc. (Cambria) performed a well recovery test on 6 May 1999 to estimate the hydraulic conductivity of the water-bearing zone beneath the site. Static water levels in wells MW-1 and MW-2 were observed to be above the screened intervals, while the water level in well MW-3 was within the screened interval. The pumping test resulted in an average of 10.5 to 11 feet of drawdown in the wells after three to four minutes of pumping at five gallons per minute. Cambria calculated the hydraulic gradient for well MW-1 to be between 9.9x10-5 centimeters per second (cm/sec) and 1.5x10-4 cm/sec. Wells MW-2 and MW-3 were calculated to be between 6.5x10-6 cm/sec and 1.7x10-5 cm/sec. The geometric mean of the hydraulic gradient for each well was calculated as 2.5x10-5 cm/sec (Cambria 2000).

2.1.6 Additional Soil & Groundwater Investigation - 2005

On 13 and 14 July 2005, URS Corporation (URS) observed the advancement of five soil borings (SB-4 through SB-8), completed by Gregg Drilling and Testing Inc. (Gregg), with the purpose of further characterizing the subsurface hydrocarbon contamination at the Site. Refer to **Figure 2** for boring locations. Borings SB-4, SB-5, and SB-7 were advanced using direct push drilling methods to a depth of 32 ft bgs, while borings SB-6 and SB-8 were advanced to a depth of 28 ft bgs. Soil samples were collected at five foot intervals from the soil borings. Hydropunch® borings were advanced on 13 and 14 July 2005, spaced one to two feet laterally from each of the five soil borings. Groundwater samples were not collected because water did not enter the boreholes. However, soil samples were collected from within the saturated zones. TPHg concentrations were detected in eleven samples collected from borings SB-4 through SB-7 at concentrations up to 1,300 mg/kg [SB-7 (2-2.5')]. Ethylbenzene was detected above laboratory reporting limits in three samples collected from borings SB-5

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

and SB-7 at concentrations up to 3.0 mg/kg [SB-7 (2-2.5')]. Total xylenes were detected in four samples collected from borings SB-6 and SB-7 at concentrations up to 3.9 mg/kg [SB-7 (5-5.5')]. Methyl tert-butyl ether (MTBE) was detected in ten samples collected from borings SB-4, SB-5, SB-6, and SB-8 at concentrations up to 3.7 mg/kg [SB-4 (29-29.5')]. Tert-butyl alcohol (TBA) was detected in two samples collected from borings SB-5 and SB-6 at concentrations up to 0.13 mg/kg [SB-6 (19.5-20')]. Other constituents analyzed for but not detected in the collected soil samples included benzene, toluene, ethanol, tert-amyl methyl ether (TAME), ethyl tert-butyl ether (ETBE), di-isopropyl ether (DIPE), 1,2-dibromoethane (EDB), and 1,2-DCA (URS 2005).

On 7 October 2005 URS observed the advancement of three off-site soil borings (SB-1, SB-2, and SB-3) and one on-site soil boring (SB-4A), completed by Gregg utilizing a combination of air knife, hand auger and hydropunch drilling methods. Refer to Figure 2 for boring locations. Off-site borings SB-1, SB-2, and SB-3 were placed between the Site and the storm drain under MacArthur Boulevard approximately one to two feet into the street from the sidewalk curb. Each offsite boring was hand augered to depth due to the proximity of underground utilities. Borings SB-1 and SB-3 were hand augered to 12 ft bgs, while boring SB-2 was hand augered to eight ft bgs. Groundwater was not encountered in the three borings, and no soil samples were collected. Boring SB-4A was placed adjacent to previous boring SB-4 to confirm subsurface soil contaminant concentrations and lithology. Boring SB-4A was advanced using direct push technology to a total depth of 36 ft bgs with groundwater first being encountered at 24.5 ft bgs. Six soil samples were collected from the boring at approximately five foot intervals. TAME was detected in one sample (SB-4A@20') at a concentration of 0.12 mg/kg. MTBE was detected in each of the six samples collected at concentrations up to 5.0 mg/kg (SB-4A@20'). The remaining analytes TPHg, BTEX, TBA, DIPE, ETBE, 1,2-DCA, EDB, and Ethanol, were below laboratory reporting limits for each of the six samples collected (URS 2006).

Also on 7 October 2005, URS observed Gregg advance four Hydropunch® borings: one each within borings SB-1, SB-2, and SB-3, and one approximately one to two feet laterally from boring SB-4A. The Hydropunch® screen was exposed in borings SB-1, SB-2, and SB-3 at 12 ft to 14 ft bgs, 14 ft to 16 ft bgs, and 17 ft to 19 ft bgs, respectively. No groundwater was encountered in these borings and therefore, no samples were collected. One groundwater sample (SB-4A) was collected from the Hydropunch® boring adjacent to boring SB-4A at a depth of 24 ft bgs. Groundwater sample SB-4A was analyzed for TPHg, BTEX, MTBE, TAME, ETBE, DIPE, TBA, EDB, 1,2-DCA, and Ethanol. TPHg was detected in the sample at a concentration of 3,000

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

• g/L, TAME at 110 • g/L, TBA at 5,700 • g/L, and MTBE at 4,500 • g/L. The remaining analytes were below the laboratory reporting limits (URS, 2006).

2.1.7 Additional Monitoring Well Installation - 2010

In September 2010 ARCADIS observed WDC Exploration & Wells (WDC) install one downgradient monitoring well (MW-4) at the location shown in **Figure 2**. Two soil samples were collected at 6.5 and 11.5 feet bgs, respectively. Soil samples were analyzed for TPHg, BTEX, MTBE, TAME, ETBE, DIPE, TBA, EDB, 1,2-DCA, and Ethanol. No concentrations were reported in the soil samples collected. Groundwater samples were collected from the well and analyzed for TPHg, BTEX, MTBE, TAME, ETBE, DIPE, TBA, EDB, 1,2-DCA, and Ethanol. No concentrations were reported in the soil samples collected. Groundwater samples were collected from the well and analyzed for TPHg, BTEX, MTBE, TAME, ETBE, DIPE, TBA, EDB, 1,2-DCA, and Ethanol. Concentrations of MTBE were reported at 95 • g/L, TBA at 6.9 • g/L and TAME at 0.75 • g/L. The remaining analytes were below the laboratory reporting limits (ARCADIS, 2010).

3. Extent of Soil and Groundwater Impacts

3.1 Soil Conditions

The vadose zone is estimated to be from grade to approximately 10 feet bgs, which is based on historical depth-to-water (DTW) readings from 1989 to present. Any soil results referred to in this section will only be for samples collected from depths not exceeding 10 feet bgs. Although saturated soil samples have been collected (at depths exceeding 10 feet bgs), it is our assumption that these concentrations may not accurately represent vadose zone soil conditions due to potential interactions with groundwater. A summary of all historical soil results is presented in **Table 1**.

Impacted soil has been encountered during removal of the waste oil USTs and associated infrastructure, and soil boring and monitoring well installation events. Based on previous investigations there appears to be two hotspots of soil contamination in the vicinity of SB-7, located in the northwestern portion of the property and TD-1 located adjacent to the northern dispenser island. The maximum concentration of all analytes detected in soil was observed in samples from SB-7, collected at 2 feet bgs (1,300 mg/kg of TPHg; 3 mg/kg ethylbenzene; 3 mg/kg of xylenes). Samples collected from 5, 9.5, and 14.5 feet bgs in SB-7 indicated 730 mg/kg, 340 mg/kg, and 0.11 mg/kg, respectively of TPHg indicating that soil contamination was isolated to shallow depths. Sample TD-1 collected under the northern pump island from 0.5 feet bgs contained 1.4 mg/kg of TPHg, 2,100 mg/Kg of TPHd and minor concentrations of fuel constituents

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

toluene (0.006 mg/kg), and total xylenes (0.04 mg/kg). Both these samples were collected in saturated soil.

ARCADIS used the Screening for Environmental Concerns at Sites with Contaminated Soil and Groundwater (RWQCB 2008) to obtain environmental screening level (ESL) values from Table A – *Shallow Soils (<3m bgs); Groundwater IS a Current or Potential Source of Drinking Water* (Water Board 2008) were used to compare environmental sample results (refer to Table 1) . The only samples to exceed the ESLs were TPHg collected from two and five feet bgs in SB-7.

The most recent soil samples taken from the vadose zone (10 feet bgs or shallower) were collected from monitoring well MW-4 in October 2010. Contaminants of concern were not detected above laboratory detection limits of from MW-4 at 6.5 feet bgs.

Historical soil data is provided in **Table 1**. The lateral extent of TPHg and MTBE soil impacts is provided on **Figures 3** and **4**, respectively. Geologic cross section locations are presented on **Figure 5** and the cross sections with the vertical extent of TPHg, benzene and MTBE soil impacts are provided on **Figures 6** and **7**.

3.2 Groundwater

Quarterly ground-water monitoring at the Site was initiated in April 1990 by Alton, and is currently performed by Broadbent and Associates Inc. (BAI). Monitoring wells MW-1, through MW-4 are currently sampled on a semi-annual basis in the first and third quarters. **Table 2** presents historical groundwater analytical data.

Historical groundwater analytical results from monitoring events are presented in **Table 2** and concentrations of TPHg and MTBE from the 2011 groundwater monitoring events are illustrated on **Figures 8** through **11**. Additionally, concentration trends for TPHg, benzene, MTBE, and TBA in wells MW-1 through MW-4 are provided in Charts 1 through 4 included in **Appendix B**.

3.2.1 TPHg

Historically TPHg has been reported at concentrations greater than laboratory reporting limits in all site wells with the exception of MW-4. The highest concentrations were reported in MW-2 (45,000 • g/L; February 2003) and MW-1 (32,000 • g/L; March 1999), which are located west and east of the USTs, respectively. TPHg has not been detected in any groundwater samples since the third quarter 2010 where MW-3

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

reported a concentration of 1,900 • g/L, which is an order of magnitude below the historical high. Concentrations of TPHg in MW-1 and MW-2 have not been reported since the third quarter 2009.

3.2.2 Benzene

Historically benzene has been reported at concentrations greater than laboratory detection limits in all site monitoring wells with the exception of MW-4. The highest concentrations were reported in MW-1 (440 \cdot g/L; May 1992), which is located east (upgradient) of the USTs. Benzene has not been detected in MW-1 since the second quarter 2009 and has decreased three orders of magnitude to its current concentration of <0.50 \cdot g/L which is below the ESL of 1.0 \cdot g/L, *Table F-1 of Revised May 2008 Screening for Environmental Concerns at Sites with Contaminated Soil and Groundwater.* Historical maximum concentrations of benzene in downgradient well MW-2 were reported at 35.10 \cdot g/L in the first quarter 2002. Benzene concentrations have not been detected in MW-2 since then.. Benzene has not been detected in any site wells since the the second quarter 2009.

3.2.3 MTBE

Concentrations of MTBE have been detected above the laboratory reporting limits in groundwater samples collected from all wells. The highest concentrations were reported in MW-1 (49,000 • g/L; March 1999) and MW-2 (39,000 • g/L; September 1999). Concentrations of MTBE in MW-1 have decreased four orders of magnitude to the current concentrations reported at 1.1 • g/L, which is below the ESLs of 5.0 • g/L. Concentrations of MTBE in MW-2 have decreased two orders of magnitude to the current concentrations reported at 170 • g/L. Concentrations of MTBE in MW-2 have decreased two orders of magnitude to the current concentrations reported at 170 • g/L. Concentrations of MTBE in MW-3 reported a historical maximum of 3,500 • g/L in the fourth quarter of 2007. The last sampling event reported a concentration of 1,700 • g/L in MW-3. MTBE concentrations in MW-4 have decreased two orders of magnitude from its historical high in February 2011 to its current concentrations reported at 3.7 • g/L which is below the ESLs of 5.0 • g/L.

3.2.4 TBA

Concentrations of TBA have been detected above the laboratory reporting limits in groundwater samples collected from all wells. The highest concentrations were reported in MW-2 (11,000 • g/L; April 2004). Concentrations of TBA in MW-2 have decreased an order of magnitude to the current concentrations reported at 2,600 • g/L.

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

Concentrations of TBA in MW-1 have decreased three orders of magnitude to the current concentrations reported at $6.5 \cdot g/L$. Concentrations of TBA in MW-3 reported a historical maximum of $200 \cdot g/L$ in the first quarter of 2006. The last sampling event reported a concentration of $110 \cdot g/L$ in MW-3. TBA concentrations in MW-4 have decreased two orders of magnitude from its historical high of $12 \cdot g/L$ in February 2011, and have not been detected in MW-4 for the last three sampling events. Currently the TBA concentrations exceed the ESLs for TBA of $12 \cdot g/L$ in MW-3. TBA concentrations in all wells have shown an overall decreasing trend in concentrations that is expected to continue.

3.3 Separate-Phase Hydrocarbon Status

Separate-phase hydrocarbon (SPH) has not been reported at the site in either soil or groundwater. Sheen has not been reported at the site in any groundwater sample.

3.4 Hydraulic Gradient Trends

Groundwater is typically encountered during drilling events from 7 feet bgs (SB-8; July 2005) to 29 feet bgs (SB-5; July 2005). The DTW in monitoring wells has ranged from 8.02 feet below top of casing (btoc) (MW-1; June 1998) to 17.97 feet btoc (MW-2; August 1991). Historically the groundwater gradient has ranged from 0.03 feet/foot to 0.09 feet/foot. The groundwater flow direction has been predominantly to the west/southwest.

Historical groundwater flow directions and gradients are provided in **Table 3** and shown on **Figure 12**. A potentiometric surface map of groundwater elevations in August 2011 is provided on **Figure 13**.

4. Beneficial Uses

4.1 San Francisco Bay RWQCB Basin Plan

Existing and potential beneficial uses for groundwater are presented in the Region 2 Water Board *San Francisco Bay Basin (Region 2) Water Quality Control Plan* (Basin Plan) dated January 18, 2007. According to the Basin Plan (Water Board 2007) the site is situated in the East Bay Plain groundwater sub-basin (basin number 2-9.04). Water supply uses including municipal, industrial process, industrial service and agricultural are identified in the Basin Plan as existing beneficial uses, based on best available information. The nearest natural drainage is Glen Echo Creek, located approximately

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

1,450 feet northwest of the Site. Glen Echo Creek flows generally northeast to southwest near the Site vicinity (BAI 2009).

4.2 Sensitive Receptor Survey

To address the potentially complete exposure pathways (groundwater, soil and soil vapor), Alisto conducted a sensitive receptor survey (SRS) in October 2000. The objective of the SRS was to identify potential downgradient and aboveground risk receptors in the vicinity of the site. Potential risk receptors included water-producing wells, schools, hospitals, surface-water bodies, and aquatic environments.

The local water supply was described as public and supplied by the East Bay Municipal Water District. The supplier's water source was said to be provided by Sierra snow melt and the Mokelumne River. Alisto identified no public water supply wells within three miles of the Site, and no private water supply wells within 0.5 mile of the site. Alisto reported that the California Department of Water Resources had no wells on record within a half mile radius of the site with the exception of the three monitoring wells associated with the site itself (Alisto 2000). However, it is known that there have since been several ground water monitoring wells installed associated with the Quick Stop/former Unocal Station No. 1871 located at the corner of MacArthur Blvd and Harrison Street. According to Conestoga Rovers and Associates' (CRA) Subsurface Investigation Report dated December 22, 2011, the current well network on the former Unocal site consists of 8 ozone sparge wells and 7 monitoring wells.

No schools are known to be located within 1,000 feet of the Site. The nearest natural drainage is Glen Echo Creek, located approximately 1,450 feet northwest of the Site. Glen Echo Creek flows generally northeast to southwest near the Site vicinity; according to the City of Oakland map it is a discontinuous creek. The nearest hospital is the Erica Breneman Kaiser Permanente Medical Center which is approximately 0.35 miles northwest (cross gradient) of the Site.

4.3 Utility Survey

In 2000 Cambria submitted a Historical Review, Utility Survey, and Recovery Testing Report which included the review of nine Sanborn fire insurance maps spanning from 1903 to 1970and ten aerial photographs spanning from 1930 to 1996. Cambria reported no visually significant historical impacts to the site or surrounding properties. Cambria also conducted a utility, or preferential pathway investigation utilizing information provided by or collected from TOSCO Corporation, Underground Service

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

Alert (USA), and a geophysical survey conducted by CU Surveys of San Ramon, California. Cambria reported that "the storm drain located beneath MacArthur Boulevard is believed to encounter groundwater at least seasonally" (Cambria 2000). A diagram of the historical utility survey is included in **Appendix C**.

4.4 Environmental Screening Levels

ARCADIS used the Revised May 2008 Screening for Environmental Concerns at Sites with Contaminated Soil and Groundwater (Water Board 2008) to obtain current ESLs and assess potential human health risks associated with current site conditions. The ESLs were developed using USEPA and California's Department of Toxic Substances Control (DTSC) human health risk assessment methodologies. Under most circumstances, the presence of a chemical in soil or groundwater at concentrations below the corresponding ESL can be assumed to not pose a significant threat to human health or the environment.

To evaluate the potential long-term fate of COCs associated with the site, the most recent and the maximum concentrations of contaminants detected in soil and groundwater have been compared to the appropriate ESLs. The use of commercial/industrial screening levels is based on the assumption that land use at the site will remain unchanged. **Figure 18** illustrates the potential sources of exposure and the status of the corresponding pathways.

Soil ESLs were obtained from *Table K-2 – Direct Contact to a Commercial/Industrial Receptor* (Water Board 2008). The ESLs are presented in **Table 4**. The most recent soil data (2010) from the vadose zone (grade to 10 feet bgs) indicates that COCs were not detected above laboratory reporting limits. The maximum detected concentrations from 2005 (1,300 mg/kg of TPHg) were collected at 2 feet bgs. Currently the entire site is paved so direct contact with soil is not possible. If future construction activities are conducted the soil should be monitored for VOCs. However, based on historical concentrations of TPHg (the only analyte above ESLs), it is unlikely elevated concentrations of TPHg would be encountered. The highest historical concentration was detected at 1,300 mg/kg in 2005 and biodegradation has occurred. Therefore the exposure through direct contact with soil is not a pathway of potential health concern.

Soil vapor samples have not been collected at the site. A station building is currently present at the site and the entire site is paved, with no exposed soil present at ground surface. The potential for soil vapor intrusion was evaluated by comparing current groundwater concentrations to groundwater ESLs obtained from Table E-1 –

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

Groundwater Screening Levels for Evaluation of Potential Vapor Intrusion Concerns (RWQCB 2008). Because groundwater does not have an active domestic use at the site, the inhalation of vapors migrating from the subsurface to indoor air is the only identified potential complete exposure pathway. Current concentrations of BTEX and MTBE are all below their applicable groundwater ESLs (Table E-1) for the protection of the indoor air pathway (**Table 4**). Concentrations of TPHg in MW-3 (1900 μ g/L) are above the groundwater ESL (29 μ g/L) for the protection of the indoor air pathway. TBA does not have an ESL for the groundwater screening levels for evaluation of potential vapor intrusion concerns. However, MW-3 is located downgradient of onsite structures and would therefore not be an issue for indoor air based on the current site configuration. Therefore, the exposure potential from groundwater through inhalation of volatile emissions is not a pathway of potential health concern.

4.5 Summary Factors Affecting Long-Term Fate of Contaminants

To evaluate the potential long-term fate of contaminants associated with the site, **Table 4** lists the results of the most recent and the maximum concentrations of contaminants detected in soil and groundwater compared to the appropriate ESLs. The use of commercial/industrial screening levels is based on the assumption that land use at the site will remain unchanged.

As shown in **Table 4**, all analytes recently sampled for soil were below their respective commercial ESLs for general soil quality with the exception of TPHd with a concentration of 2,100 mg/Kg at sample location TD-1. This sample was collected in 1994 from beneath a dispenser island. Additionally concentrations of TPHg were detected in samples collected from two (1,300 mg/kg) and five (730 mg/kg) feet bgs from SB-7 in 2005 (**Table 1**). The entire site is paved and the commercial/industrial worker would not come into contact with the soil. Therefore, the Site, under current or future commercial use, does not pose as a human health risk to the hypothetical onsite worker populations.

The most recent TPHg concentrations observed in a groundwater sample from MW-1 was $120 \cdot g/L$ in September of 2009. Concentrations of TPHg in MW-1 have indicated decreasing trends since 1998 and are expected to continue to decline. The most recent TPHg concentrations observed at MW-2 and MW-3 were reported at less than the detection limits of $250 \cdot g/L$ and $1,000 \cdot g/L$, respectively in February 2012. MW-2 has maintained a steadily declining trend of TPHg since 2003 and is expected to continue. The most recent TPHg concentrations observed at well MW-3 reported a detection of 1,900 $\cdot g/L$ in August 2010. Historically, concentration trends at MW-1 and MW-2 for

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

TPHg increased, fluctuated, and finally decreased to below (MW-1) or near ESLs (MW-2). MW-3 is located down/cross gradient of these wells and it is expected the concentrations will mimic this pattern. Currently concentrations are fluctuating and beginning to decrease. The concentrations in MW-1 and MW-2 were an order of magnitude higher than MW-3 so it is expected the ESL for the protection of the indoor air pathway will be reached in the near future. Concentrations of MTBE in MW-3 have continued to decline over its history to their current concentrations that are below the ESLs (**Appendix B**).

The most recent MTBE concentrations observed at well MW-1 reported a detection of 14 • g/L in February 2012. Concentrations of MTBE in MW-1 have maintained an overall decreasing trend which is expected to continue. The most recent MTBE concentrations observed at well MW-2 reported a detection of 300 • g/L in February 2012. Concentrations of MTBE in MW-2 have maintained an overall decreasing trend which is expected to continue. The most recent MTBE concentrations observed at well MW-2 have maintained an overall decreasing trend which is expected to continue. The most recent MTBE concentrations observed at well MW-3 reported a detection of 1,700 • g/L in February 2012. Concentrations of MTBE in MW-3 have recently exhibited an overall decreasing trend. Based on trend similarities between MW-1, MW-2 and MW-3 this trend is expected to continue. The most recent MTBE concentrations observed at well MW-4 reported a detection of 12 • g/L in February 2012. Concentrations of MTBE in MW-4 have maintained an overall decreasing trend which is expected to continue. The most recent MTBE concentrations of MTBE in MW-4 have maintained an overall decreasing trend which is expected to continue (**Appendix B**). As shown in **Table 4**, MTBE concentrations are below the ESL for the protection of the indoor air pathway.

The most recent TBA concentrations observed at well MW-2 reported a detection of $2,600 \cdot g/L$ in February 2012. Concentrations of TBA in MW-2 have maintained an overall decreasing trend which is expected to continue. The most recent TBA concentrations observed at well MW-3 reported a detection of $110 \cdot g/L$ in February 2012. Concentrations of TBA in MW-3 have maintained an overall decreasing trend which is expected to continue.

The historical maximum groundwater concentrations of MTBE from MW-2 were reported at 35,000 • g/L and the maximum historical MTBE concentration in MW-3 was reported at 3,500 • g/L. The location of MW-3 is approximately 70 feet north (down/cross gradient) of MW-2 and shows an order of magnitude reduction in MTBE concentrations in that distance. MW-4 is located approximately 70 feet down/cross gradient of MW-3 and contains two orders of magnitude less than the concentrations at MW-3. Based on the degradation of contaminants approximately one order of magnitude every 70 feet the contaminant plume is not expected to extend beyond the

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

underpass of Highway 580. The MTBE historic concentrations are also below the ESL for the protection of the indoor air pathway.

5. Remedial Activities

The potential source of hydrocarbons includes the UST complex in the southern portion of the Site, and the dispenser islands. The exact volume released from the UST complex and product piping are unknown.

The removal of the original waste oil UST in 1988 and UST and dispenser replacement 1990 was conducted as a release intervention. During removal of the waste oil UST approximately 15 cubic yards of contaminated soil was excavated and removed at the time of the waste oil UST removal (BAI, 2009).

Numerous soil borings and monitoring wells have been installed to delineate and monitor the lateral and vertical extent of petroleum hydrocarbon impacts. Remediation through site upgrades, equipment removal and associated over-excavations, and natural attenuation have proven to be effective for substantially removing on-site contamination sources. It has been demonstrated by declining petroleum hydrocarbon concentration trends in site monitoring wells that natural attenuation is occurring and should continue to occur at the site.

6. Remedial Effectiveness

Based upon the previous remedial activities detailed above, the impacts to soil and groundwater have been addressed and are evident by the low levels of soil impacts and decreasing concentration trends in groundwater. Downgradient groundwater samples from monitoring well MW-4 have largely been below laboratory detection limits since its installation in October 2010 with the exception of low level MTBE detections (**Table 2/ Appendix B**). The lack of concentrations seen in the downgradient wells indicates that any residual groundwater impacts are not migrating off site.

7. Conclusions

The site meets all published criteria and qualifies as low risk, as described in the *Supplemental Instructions to State Water Board December 18, 1995 Interim Guidance of Required Cleanup at Low-Risk Fuel Sites* (Water Board 1996). Therefore, ARCADIS requests approval for case closure and no further action at this site based on the following:

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

- Sources of petroleum hydrocarbons at the site have been adequately addressed. The absence of high concentrations of site COCs observed in soil samples suggests that secondary sources (residual hydrocarbons in soil) were identified and removed.
- Concentrations of TPHg are all below reporting limits.
- All the wells that contain MTBE concentrations in groundwater indicate a decreasing trend (Appendix D).
- All the wells that contain TBA concentrations in groundwater indicate a decreasing trend (**Appendix D**).
- Concentrations of BTEX in groundwater are all below the reporting limits.
- Current site conditions suggest that soil concentrations of TPHg are limited to the western portion of the site in the vicinity of SB-7 (**Figure 7**).
- Current site conditions suggest that soil concentrations of MTBE are limited to the western portion of the site in the vicinity of SB-7 (**Figure 7**).
- The plume does not appear to be migrating offsite, as evidenced by the results of groundwater samples collected in MW-4.
- The site has been adequately characterized.
- No sensitive receptors are likely to be impacted, including surface-water bodies, municipal wells and drinking water sources.
- The site presents no current or potential risk to human health or the environment.

ARCADIS recommends that case closure be granted and that all groundwater monitoring wells associated with the site be destroyed. A Case Closure Summary is included in **Appendix D**.

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

8. References

ACEH. 2009. Fuel Leak Case No. RO0000456 and GeoTracker Global ID T0600100908, BP #11102, 100 W. MacArthur Boulevard, Oakland, CA 94610. Submitted to Mr. Paul Supple for Atlantic Richfield Company, by Mr. Paresh Khatri. January 8.

Alisto Engineering Group, 2000. Potential Receptor Survey, Expanded Site Plan and Well Search, BP Oil Company Service Station No. 11102, 100 MacArthur Boulevard, Oakland, California. 19 October.

Alton Geoscience, Inc., 1989. Preliminary Site Investigation Report, Former Mobil Service Station No. 10-E6A, 100 MacArthur Boulevard, Oakland, California. 20 December

Alton Geoscience, Inc., 1990. Quarterly Ground Water Monitoring and Sampling Report for Former Mobil Service Station 10-E6A, 100 MacArthur Boulevard, Oakland, California. 2 May.

ARCADIS U.S., 2010. Monitoring Well Installation Report, Former BP Service Station No. 11102, 100 MacArthur Boulevard, Oakland, California, ACEH Case #RO0000456. 30 November.

Broadbent & Associates, Inc. 2009. Initial Site Conceptual Model with Soil and Groundwater Investigation Work Plan. Former BP Service Station No. 11102, 100 MacArthur Boulevard, Oakland, California. 9 April.

California Regional Water Quality Board, San Francisco Bay Region, Groundwater Committee. 1999. East Bay Plain Groundwater Basin Beneficial Use Evaluation Report, Alameda and Contra Costa Counties, CA. June.

Cambria Environmental Technology, Inc., 2000. Historical Review, Utility Survey and Recovery Testing Report, BP Oil Site No. 11102, 100 MacArthur Boulevard, Oakland, California. 24 February.

Environmental Resolutions, Inc., 1998. Work Plan for Environmental Work at Tosco BP Service Station #11102, 100 MacArthur Boulevard, Oakland, California. 11 July.

Recommendation for Case Closure

Former Atlantic Richfield Company Station No. 11102

Kaprealian Engineering, Inc., 1988. Soil Sampling Report, Mobil Service Station #10-E6A, 100 MacArthur Blvd., Oakland, California. 7 October.

Regional Water Quality Control Board (RWQCB). 2008. Screening for Environmental Concerns at Sites with Contaminated Soil and Groundwater. Interim Final - November 2007. Revised May 2008.

URS Corporation, 2005. Soil and Water Investigation Report, Former BP Service Station #11102, 100 MacArthur Boulevard, Oakland, California, ACEHS Case No. RO0000456. 4 August.

URS Corporation, 2006. Supplemental Soil and Water Investigation Report, Former BP Service Station # 11102, 100 MacArthur Boulevard, Oakland, California, ACEHS Case No. RO0000456. 14 April.

Tables

| Location | Sample Depth Sample Date TPHg (ft bgs) | | Hg | TPHd Benzene | | Toluene Ethy | | Ethylbenzene Xyle | | ene MTBE | | BE | ТВА | | Total O & G | | Lead | | | | | |
|------------------------|--|--|----------------------|-------------------------|---------------------|-------------------------|----------------------------|-------------------------|----------------------------|-------------------------|----------------------------|-------------------------|----------------------------|-------------------------|-----------------------|-------------------------|-----------------------|-------------------------|-------------------|-------------------------|----------|----------|
| Commerc | cial ESLs | (mg/Kg) ¹ | 45 | 450 | | 50 | 0.27 | | 21 | 0 | 5.0 | 0 | 10 | 0 | 65 | 5 | - | | - | - | | |
| WO Comp WO | 9 | 9/19/1988 9/19/1988 | | | 2.0 1,700 | mg/Kg mg/Kg | <100 | µg/Kg | <100 | µg/Kg | <100 | µg/Kg | <100 | µg/Kg | | | | | 24 65,000 | mg/Kg mg/Kg | | |
| MW-1 MW-1 MW-1 | 5 10 15 | 10/26/1989 10/26/1989 10/26/1989 | | | <10 <10 <10 | mg/Kg mg/Kg mg/Kg | <0.005 <0.005 <0.005 | mg/Kg mg/Kg mg/Kg | <0.005 <0.005 <0.005 | mg/Kg mg/Kg mg/Kg | <0.005 <0.005 <0.005 | mg/Kg mg/Kg mg/Kg | <0.005 <0.005 <0.005 | mg/Kg mg/Kg mg/Kg | | | | | <30 <30 <30 | mg/Kg mg/Kg mg/Kg | | |
| MW-2 MW-2 MW-2 | 5 10 15 | 10/25/1989 10/25/1989 10/25/1989 | <1.0 <1.0 <1.0 | mg/Kg mg/Kg mg/Kg | | | 0.006 0.008 <0.005 | mg/Kg mg/Kg mg/Kg | <0.005 <0.005 <0.005 | mg/Kg mg/Kg mg/Kg | <0.005 <0.005 <0.005 | mg/Kg mg/Kg mg/Kg | <0.005 <0.005 <0.005 | mg/Kg mg/Kg mg/Kg | | | | | | | | |
| MW-3 MW-3 MW-3 | 5 10 15 | 10/26/1989 10/26/1989 10/26/1989 | <1.0 <1.0 <1.0 | mg/Kg mg/Kg mg/Kg | | | <0.005 <0.005 <0.005 | mg/Kg mg/Kg mg/Kg | 0.006 <0.005 <0.005 | mg/Kg mg/Kg mg/Kg | <0.005 <0.005 <0.005 | mg/Kg mg/Kg mg/Kg | 0.013 <0.005 <0.005 | mg/Kg mg/Kg mg/Kg | | | | | | | | |
| MW-4 MW-4 | 6.5 11.5 | 10/6/2010 10/6/2010 | <0.3 <0.3 | mg/Kg mg/Kg | | | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | <0.01 <0.01 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | <0.01 <0.01 | mg/Kg mg/Kg | | | | |
| TD-1 TD-3 | 0.5 0.5 | 11/22/1994 11/22/1994 | 1.4 <1.0 | mg/Kg mg/Kg | 2,100 470 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | 0.006 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | 0.04 <0.005 | mg/Kg mg/Kg | | | | | <1.0 <1.0 | mg/Kg mg/Kg | | |
| THP-1 THP-1 | 13 15.5 | 11/22/1994 11/22/1994 | 1.2 <1.0 | mg/Kg mg/Kg | <1.0 <1.0 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | | | | | <1.0 <1.0 | mg/Kg mg/Kg | | |
| THP-2 THP-2 | 7 10 | 11/22/1994 11/22/1994 | <1.0 <1.0 | mg/Kg mg/Kg | <1.0 <1.0 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | | | | | <1.0 <1.0 | mg/Kg mg/Kg | | |
| SB-1 SB-2 SB-3 | | 10/7/2005 10/7/2005 10/7/2005 | | | | | | | | | | | | | | | | | | | | |
| SB-4 SB-4 SB-4 | 5 9.5 14.5 | 7/14/2005 7/14/2005 7/14/2005 | <1.0 <0.50 | mg/Kg mg/Kg mg/Kg | | | <0.005 <0.025 <0.050 | mg/Kg mg/Kg mg/Kg | <0.005 <0.025 <0.050 | mg/Kg mg/Kg mg/Kg | <0.005 <0.025 | mg/Kg mg/Kg mg/Kg | <0.005 <0.025 | mg/Kg mg/Kg mg/Kg | <0.005 0.37 1 1 | mg/Kg mg/Kg mg/Kg | <0.020 <0.10 | mg/Kg mg/Kg mg/Kg | | | | |
| SB-4 SB-4 SB-4 | 19.5 20 25 | 7/14/2005 7/14/2005 7/14/2005 | 3.8 <12 <25 | mg/Kg mg/Kg mg/Kg | | | <0.050 <0.25 <0.50 | mg/Kg mg/Kg mg/Kg | <0.050 <0.25 <0.50 | mg/Kg mg/Kg mg/Kg | <0.050 <0.25 <0.50 | mg/Kg mg/Kg mg/Kg | <0.050 <0.25 <0.50 | mg/Kg mg/Kg mg/Kg | 2.4 3.4 3.5 | mg/Kg mg/Kg mg/Kg | <5.0 <25 <25 | mg/Kg mg/Kg mg/Kg | | | | |
| SB-4 SB-4A SB-4A | 29 6 10 | 7/14/2005 10/7/2005 10/7/2005 | <25 <0.25 <2.5 | mg/Kg mg/Kg mg/Ka | | - | <0.50 <0.012 <0.050 | mg/Kg mg/Kg ma/Ka | <0.50 <0.012 <0.050 | mg/Kg mg/Kg mg/Ka | <0.50 <0.012 <0.050 | mg/Kg mg/Kg mg/Ka | <0.50 <0.012 <0.050 | mg/Kg mg/Kg mg/Ka | 3.7 0.073 0.2 | mg/Kg mg/Kg mg/Ka | <50 <0.050 <5.0 | mg/Kg mg/Kg ma/Ka | - | - | | |
| SB-4A | 20 | 10/7/2005 | <5.0 | mg/Kg | | | <0.10 | mg/Kg | <0.10 | mg/Kg | <0.10 | mg/Kg | <0.10 | mg/Kg | 5.0 | mg/Kg | <10 | mg/Kg | | | | |
| SB-4A SB-4A | 25 30 | 10/7/2005 | <2.5 <0.010 | mg/Kg ma/Ka | | | <0.050 <0.0050 | mg/Kg ma/Ka | <0.050 <0.0050 | mg/Kg ma/Ka | <0.050 <0.0050 | mg/Kg ma/Ka | <0.050 <0.0050 | mg/Kg ma/Ka | 0.84 | mg/Kg ma/Ka | <5.0 <0.020 | mg/Kg ma/Ka | | | | |
| SB-4A SB-5 | 35 5 | 10/7/2005 7/14/2005 | <0.10 <0.099 | mg/Kg mg/Kg | | | <0.0050 <0.005 | mg/Kg mg/Kg | <0.0050 <0.005 | mg/Kg mg/Kg | <0.0050 <0.005 | mg/Kg mg/Kg | <0.0050 <0.005 | mg/Kg mg/Kg | 0.057 <0.005 | mg/Kg mg/Kg | <0.020 | mg/Kg mg/Kg | | | | |
| SB-5 SB-5 | 9.5 14.5 | 7/14/2005 | 0.15 | mg/Kg | | | < 0.005 | mg/Kg mg/Kg | < 0.005 | mg/Kg | < 0.005 | mg/Kg | < 0.005 | mg/Kg | < 0.005 | mg/Kg | <0.020 | mg/Kg | | | | |
| SB-5 | 19.5 | 7/14/2005 | 61 | mg/Kg | | | <0.005 | mg/Kg | <0.005 | mg/Kg | 0.14 | mg/Kg | <0.005 | mg/Kg | <0.005 | mg/Kg | <5.0 | mg/Kg | | - | | |
| SB-5 SB-6 | 29 5 | 7/14/2005 7/13/2005 | 0.1 <0.10 | mg/Kg mg/Kg | | | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | 0.053 <0.005 | mg/Kg mg/Kg | 0.65 <0.020 | mg/Kg mg/Kg | | - | | |
| SB-6 | 8.5 | 7/13/2005 | <0.10 | mg/Kg | | | <0.005 | mg/Kg mg/Kg | <0.005 | mg/Kg mg/Kg | <0.005 | mg/Kg | <0.005 | mg/Kg | <0.005 | mg/Kg | <0.020 | mg/Kg | | | | |
| SB-6 | 14.5 | 7/13/2005 | <0.097 | mg/Kg | | | <0.0048 | mg/Kg | <0.0048 | mg/Kg | <0.0048 | mg/Kg | 0.0082 | mg/Kg | <0.0048 | mg/Kg | <0.019 | mg/Kg | | | | |
| SB-6 | 16.5 | 7/13/2005 | < 0.098 | mg/Kg | | | < 0.0049 | mg/Kg | < 0.0049 | mg/Kg | < 0.0049 | mg/Kg | 0.0054 | mg/Kg | < 0.0049 | mg/Kg | < 0.020 | mg/Kg | | | | |
| SB-6 | 27.5 | 7/13/2005 | <0.50 | mg/Kg | | - | <0.025 | mg/Kg | <0.025 | mg/Kg | <0.025 | mg/Kg | <0.025 | mg/Kg mg/Kg | <0.005 | mg/Kg mg/Kg | <0.020 | mg/Kg mg/Kg | | | | |
| SB-7 SB-7 | 2 | 7/14/2005 7/14/2005 | 1300 730 | mg/Kg mg/Ka | | - | <1.0 <1.0 | mg/Kg mg/Ka | <1.0 <1.0 | mg/Kg mg/Kg | 3.0 2.4 | mg/Kg mg/Ka | 3.0 3.9 | mg/Kg mg/Ka | <0.50 <0.50 | mg/Kg mg/Ka | <100 <100 | mg/Kg ma/Ka | | | | |
| SB-7 | 9.5 | 7/14/2005 | 340 | mg/Kg | | | <2.5 | mg/Kg | <2.5 | mg/Kg | <2.5 | mg/Kg | <2.5 | mg/Kg | <1.2 | mg/Kg | <250 | mg/Kg | | | | |
| SB-7 | 14.5 | 7/14/2005 | 0.11 | mg/Kg | | | < 0.005 | mg/Kg | < 0.005 | mg/Kg | < 0.020 | mg/Kg | | | | |
| 58-7 SB-7 | 19.5 25.5 | 7/14/2005 | <0.099 | mg/Kg mg/Kg | | | <0.005 | mg/Kg mg/Kg | <0.005 | mg/Kg mg/Kg | <0.005 | mg/Kg mg/Kg | <0.005 | mg/Kg mg/Kg | <0.005 | mg/Kg mg/Kg | <0.020 | ng/Kg ma/Ka | | - | - | |
| SB-7 | 28.5 | 7/14/2005 | <0.10 | mg/Kg | | | <0.005 | mg/Kg | <0.005 | mg/Kg | <0.005 | mg/Kg | <0.005 | mg/Kg | <0.005 | mg/Kg | <0.020 | mg/Kg | | | | |
| SB-7 SB-8 | 30.5 5 | 7/14/2005 7/13/2005 | <0.10 | mg/Kg mg/Ka | | - | <0.005 <0.005 | mg/Kg mg/Ka | <0.005 | mg/Kg mg/Ka | <0.005 <0.005 | mg/Kg mg/Ka | <0.005 <0.005 | mg/Kg mg/Ka | <0.005 <0.005 | mg/Kg mg/Ka | <0.020 <0.20 | mg/Kg mg/Ka | | | | |
| SB-8 | 7 | 7/13/2005 | <0.10 | mg/Kg | | | <0.005 | mg/Kg | <0.005 | mg/Kg | < 0.005 | mg/Kg | <0.005 | mg/Kg | <0.005 | mg/Kg | <0.20 | mg/Kg | | | | |
| SB-8 SB-8 | 9.5 11 | 7/13/2005 | <0.099 | mg/Kg | | - | <0.005 | mg/Kg | <0.005 | mg/Kg | <0.005 | mg/Kg | <0.005 | mg/Kg | <0.005 | mg/Kg | <0.20 | mg/Kg | | | | |
| SB-8 | 14.5 | 7/13/2005 | <0.099 | mg/Kq | - | - | < 0.005 | mg/Kq | < 0.005 | mg/Kq | <0.005 | mg/Kq | <0.005 | mg/Kq | <0.005 | mg/Kq | <0.20 | mg/Kg | | | | |
| SB-8 | 17.5 | 7/13/2005 | <0.10 | mg/Kg | | | < 0.005 | mg/Kg | < 0.005 | mg/Kg | <0.20 | mg/Kg | | | | |
| SB-8 SB-8 | 19.5 20.5 | 7/13/2005 7/13/2005 | <0.10 <0.10 | mg/Kg mg/Kg | | | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | <0.005 <0.005 | mg/Kg mg/Kg | 0.066 | mg/Kg mg/Kg | <0.20 <0.20 | mg/Kg mg/Kg | | | | |

Notes: ft bgs = feet below ground surface TPHg = Total Petroleum Hydrocarbons as Gasoline TPHd = Total Petroleum Hydrocarbons as Diesel mg/kg = milligrams per kilogram ppm = parts per million MTBE = Methyl tert-butyl ether - = not analyzed < = analyte not detected, result is less than value provided SB-1, SB-2 and SB-3 not sampled for soil 1 = Soil ESLs values are listed from Table K-2

| Location | Sample Date | Notes | TOC Elevation (ft) | DTW (ft btoc) | Product Thickness (ft) | Water Level Elevation | TPHg | Benzene | Toluene | Ethylbenzen e | Xylene | МТВЕ | ТВА | TPHd |
|----------|-----------------------|-------|--------------------------|------------------|------------------------------|-----------------------------|------------|----------|--------------|------------------|-------------|------------|-----------|--------------|
| | | | (, | | () | (ft) | | | | μg | J/L | | | |
| MW-1 | 11/4/1989 | GW | 90.20 | 13.21 | | 76.99 | <500 | 3.4 | 0.6 | <0.3 | <0.3 | | | <50 |
| | 4/3/1990 7/20/1000 | GW | 90.20 | 12.40 | | 77.74 | 820 100 | 64 11 | 1.9 | 23 | 34 -5 | | | -50 |
| | 11/20/1000 | GW | 90.20 | 14.09 | | 77.28 | 50 | 24 | <0.2 | <0.2 | <0.2 | | | <00 70 |
| | 3/1/1001 | GW | 90.20 | 13.61 | | 76.12 | ~100 | 2.4 | <0.3 | <0.3 | <0.3 0.3 | | | /9 <1.000 |
| | 8/10/1001 | GW | 90.20 | 15.01 | | 76.59 | 370 | 0.9 | 0.73 | <0.3 6.4 | 0.3 | | | < 50 |
| | 0/19/1991 | GW | 90.20 | 14.09 | | 74.46 | 570 | 0.69 | -0.2 | -0.2 | -0.2 | | | <50 |
| | 2/24/1002 | GW | 90.20 | 12.52 | | 76.12 | 140 | 3.0 | <0.3 0.66 | 1.2 | 3.8 | | | 100 |
| | 5/10/1002 | GW | 90.20 | 11.80 | | 79.40 | 4 200 | 440 | 21 | 250 | 37 | | | 910 |
| | 6/17/1992 | GW | 90.20 | 12.01 | | 78.40 | 4,200 | 350 | 14 | 150 | 17 | | | 560 |
| | 7/22/1992 | GW | 90.20 | 12.01 | | 70.19 | 4 000 | <5 | 19 | 210 | 61 | | | 000 |
| | 8/14/1992 | GW | 90.20 | 12 75 | | 77.45 | 2 400 | 330 | 20 | 150 | 47 | | | 1 700 |
| | 11/11/1992 | GW | 90.20 | 13.69 | | 76.51 | 260 | 30 | 3.4 | 7.6 | 6.8 | | | 92 |
| | 6/7/1993 | GW | 90.20 | 10.93 | | 79.27 | 3.400 | 98 | 11 | 21 | 7.6 | | | 440 |
| | 12/2/1993 | GW | 90.20 | 12.72 | | 77.48 | 1,100 | 8.3 | 3.6 | 0.6 | 1.5 | | | 120 |
| | 6/22/1994 | GW | 90.20 | 11.81 | | 78.39 | 2,100 | 32 | 3.8 | 2.2 | 17 | 4,000 | | <50 |
| | 1/10/1995 | GW | 90.20 | 10.97 | | 79.23 | <500 | 120 | <5 | <5 | <10 | , | | 420 |
| | 6/21/1995 | GW | 90.20 | 9.38 | | 80.82 | 4,700 | 16 | <5 | <5 | <10 | | | 1,300 |
| | 12/27/1995 | GW | 90.20 | 11.55 | | 78.65 | 430 | <2.5 | <2.5 | <2.5 | <5 | 1,200 | | 2,100 |
| | 6/13/1996 | GW | 90.20 | 9.28 | | 80.92 | 3,200 | 51 | <12 | <12 | <12 | 4,000 | | 920 |
| | 12/4/1996 | GW | 90.20 | 11.91 | | 78.29 | 1,400 | 6.2 | <5 | <5 | <5 | 2,600 | | 280 |
| | 6/10/1997 | GW | 90.20 | 8.97 | | 81.23 | 7,900 | 12 | <10 | <10 | <10 | 15,000 | | 1,700 |
| | 12/12/1997 | GW | 90.20 | 11.37 | | 78.83 | 440 | 8.8 | <1 | 2.6 | 9.4 | 6,700 | | 760 |
| | 6/18/1998 | GW | 90.20 | 8.02 | | 82.18 | 7,500 | <2.5 | <5 | <5 | <5 | 5,600 | | 2,900 |
| | 3/9/1999 | GW | 90.20 | 9.80 | | 80.40 | 32,000 | 100 | 16 | 72 | 110 | 49,000 | | |
| | 9/28/1999 | GW | 90.20 | 10.78 | | 79.42 | 1,000 | <5 | <5 | <5 | <5 | 730 | | |
| | 10/14/1999 | GW | 90.20 | 10.84 | | 79.36 | | | | | | | | 660 |
| | 3/27/2000 | GW | 90.20 | 9.83 | | 80.37 | 4,300 | 160 | 19 | 37 | 43 | 28,000 | | |
| | 9/28/2000 | GW | 90.20 | 11.33 | | 78.87 | 2,700 | 10 | 2.6 | 1.1 | 2.7 | 28,000 | | |
| | 3/8/2001 | GW | 90.20 | 10.96 | | 79.24 | 8,200 | 23.5 | 6.09 | 5.23 | 8.97 | 11,600 | | |
| | 9/21/2001 | GW | 90.20 | 12.07 | | 78.13 | 6,000 | 37.9 | <0.5 | <0.5 | <1.5 | 7,370 | | |
| | 2/28/2002 | GW | 90.20 | 10.48 | | 79.72 | 6,400 | 60.8 | <5 | 6.43 | <10 | 7,750 | | |
| | 9/6/2002 | GW | 90.20 | 11.20 | | 79.00 | 1,400 | <5 | <5 | <5 | <5 | 6,000 | | |
| | 2/19/2003 | GW | 90.20 | 11.29 | | 78.91 | <10,000 | <100 | 110 | <100 | <100 | 4,500 | | |
| | 7/14/2003 | GW | 90.20 | 11.18 | | 79.02 | 710 | 11 | <10 | <10 | <10 | 940 | 2,700 | |
| | 1/14/2004 | GW | 90.20 | 11.74 | | 78.46 | <500 | <5 | <5 | <5 | <5 | 220 | 2,500 | |
| | 4/23/2004 | GW | 90.20 | 11.95 | | 78.25 | 470 | 3.4 | <2.5 | <2.5 | <2.5 | 150 | 2,500 | |
| | 7/1/2004 | GW | 90.20 | 11.52 | | 78.68 | 360 | <2.5 | <2.5 | <2.5 | <2.5 | 96 | 2,000 | |
| | 10/28/2004 | GW | 90.20 | 12.56 | | 77.64 | 390 | 0.94 | <0.5 | <0.5 | <0.5 | 43 | 1,500 | |
| | 1/10/2005 | GW | 90.20 | 11.85 | | 78.35 | 490 | 17 | <2.5 | 5.8 | 5.4 | 85 | 1,900 | |
| | 4/13/2005 | GW | 90.20 | 10.00 | | 80.20 | 1,000 | 27 | <2.5 | <2.5 | 25 | 48 | 1,400 | |
| | 7/11/2005 | GW | 90.20 | 9.27 | | 80.93 | 180 | <0.5 | <0.5 | <0.5 | <0.5 | 36 | 550 | |
| | 10/17/2005 | GW | 90.20 | 10.96 | | 79.24 | 140 | <0.5 | <0.5 | <0.5 | <0.5 | 20 | 450 | |
| | 1/17/2006 | GW | 90.20 | 10.81 | | 79.39 | 120 | 0.64 | <0.5 | <0.5 | 0.56 | 38 | 260 | |
| | 4/21/2006 | GW | 90.20 | 9.28 | | 80.92 | 410 | 1.4 | 1 | <0.5 | <0.5 | 17 | 320 | |
| | 7/17/2006 | GW | 90.20 | 9.25 | | 80.95 | <50 | <0.5 | <0.5 | <0.5 | <0.5 | 5.5 | 32 | |
| | 7/26/2006 | GW | 90.20 | 8.57 | | 81.63 | <50 | <0.5 | <0.5 | <0.5 | <0.5 | 4.4 | 22 | |
| | 10/31/2006 | GW | 90.20 | 9.80 | | 80.40 | <50 | <0.5 | <0.5 | <0.5 | <0.5 | 2.8 | <20 | |
| | 1/8/2007 | GW | 90.20 | 10.36 | | 79.84 | <50 | 2.2 | <0.5 | <0.5 | <0.5 | 6.2 | 110 | |
| | 4/10/2007 | GW | 90.20 | 10.65 | | 79.55 | 100 | 1.4 | <0.5 | <0.5 | <0.5 | 9 | 210 | 400 |
| | 7/10/2007 | GW | 90.20 | 10.52 | | 79.68 | 120 | <0.5 | <0.5 | <0.5 | <0.5 | 4.9 | 110 | 160 |
| | 1/22/2000 | GW | 90.20 | 11.23 | | 78.97 | 2/0 | <0.5 | <0.5 | <0.5 | <0.5 | 4.9 | 94 110 | |
| | 1/22/2008 | GW | 90.20 | 10.26 | | 78.98 | 240 | <0.5 | <0.5 | 0.03 | 0.72 | 1.Z E E | 04 | |
| | 4/10/2008 | GW | 90.20 | 10.20 | | 79.94 | 240 79 | <0.5 | <0.5 | <0.5 | 0.73 | 5.5 5.9 | 04 64 | |
| | 11/10/2000 | GW | 90.20 | 12 51 | | 79.10 | 150 | <0.5 | <0.5 | <0.0 >0 F | <0.5 | 3.0 | 110 | |
| | 2/10/2000 | GW | 90.20 | 12.01 | | 77.40 | 150 | <0.5 | <0.5 | <0.0 -0 F | <0.5 | 5.4 | 110 | |
| | 5/7/2009 | GW | 90.20 | 10.00 | | 77.49 | <50 | <0.5 | <0.5 | <0.5 | <0.5 | 5.3 12 | 17 | |
| | 9/3/2009 | GW | 90.20 | 11 01 | | 79.30 | 120 | -0.5 | <0.5 | <0.5 | 0.0 | 3.2 | 260 | |
| | 10/20/2009 | GW | 90.20 | 12.54 | | 10.29 | <50 | <0.5 | <0.5 | <0.5 | 0.09 ~1 | 22 | 200 | |
| | 2/26/2010 | GW | 90.20 | 10.61 | | 70.50 | <50 | <0.5 | <0.5 | ~0.5 | -1 | R 1 | 240 | |
| | 8/16/2010 | GW | 90.20 | 10.01 | | 1 9.09 80 00 | <50 | <0.5 | <0.5 | <0.5 | -1 | 8.1 | 35 | |
| | 11/12/2010 | 0,0 | 90.20 | 10.53 | | 79.67 | -00 | -0.0 | -0.0 | -0.0 | | 0.1 | 00 | |
| 1 | 1 | I | 00.20 | 1 | I | | I | I | I | I | I . | I | 1 | ı I |

| Location | Sample Date | Notes | TOC Elevation (ft) | DTW (ft btoc) | Product Thickness (ft) | Water Level Elevation | TPHg | Benzene | Toluene | Ethylbenzen e | Xylene | MTBE | ТВА | TPHd |
|----------|----------------|-------|--------------------------|------------------|------------------------------|-----------------------------|-----------------|--------------|-------------|------------------|-------------|--------|---------|------|
| | 2/2/2011 | 0.14 | 00.00 | 11.00 | | (11) | -50 | -0 F | -0 F | P9 | /∟ 1 | 14 | -250 | |
| | 2/3/2011 | GW | 90.20 | 11.88 | | 78.32 | <50 | <0.5 | <0.5 | <0.5 | <1 | 14 | <250 | |
| | 6/23/2011 | | 90.20 | 9.78 | | 80.42 | | | | | | | | |
| | 8/22/2011 | GW | 90.20 | 10.39 | | 79.81 | <50 | <0.5 | <0.5 | <0.5 | <1 | 1.1 | <250 | |
| | 2/20/2012 | GW | 90.20 | 11.53 | | 78.67 | <50 | <0.5 | <0.5 | <0.5 | <1 | 14 | 6.5 | |
| MW-2 | 11/4/1989 | GW | 87.91 | 15.84 | | 72.07 | <500 | 6.5 | <0.3 | <0.3 | <0.3 | | | |
| | 4/3/1990 | GW | 87.91 | 15.25 | | 72.66 | <500 | <0.5 | <0.5 | <0.5 | <0.5 | | | |
| | 7/30/1990 | GW | 87.91 | 15.59 | | 72.32 | 61 | 6.5 | <0.5 | <0.5 | < 0.5 | | | |
| | 11/20/1990 | GW | 87.91 | 17.81 | | 70.10 | <50 | 0.3 | <0.3 | <0.3 | <0.3 | | | |
| | 8/10/1001 | GW | 07.91 87.01 | 17.11 | | 70.80 60.94 | <100 | -0.4 ∠0.3 | <0.3 | <0.3 | <0.3 | | | |
| | 11/13/1991 | GW | 87.91 | 16.76 | | 71 15 | 38 | 0.32 | <0.3 | <0.3 | <0.3 | | | |
| | 2/24/1992 | GW | 87.91 | 15.07 | | 72.84 | <50 | <0.5 | <0.5 | <0.5 | 0.58 | | | |
| | 5/19/1992 | GW | 87.91 | 14.70 | | 73.21 | <50 | 0.55 | <0.5 | <0.5 | <0.5 | | | |
| | 7/22/1992 | GW | 87.91 | 15.60 | | 72.31 | 90 | 1.3 | 0.6 | 0.9 | 1.9 | | | |
| | 11/11/1992 | GW | 87.91 | 16.19 | | 71.72 | 52 | 2.8 | <0.5 | <0.5 | 0.9 | | | |
| | 6/7/1993 | GW | 87.91 | 14.42 | | 73.49 | 1,200 | 14 | 2.8 | 1.9 | 1.71 | | | |
| | 12/2/1993 | GW | 87.91 | 14.94 | | 72.97 | 790 | 3.4 | 0.5 | 10 | < 0.5 | 3,700 | | |
| | 6/22/1994 | GW | 87.91 | 14.25 | | 73.66 | 110 | <0.5 | <0.5 | <0.5 | <0.5 | 120 | | |
| | 6/21/1005 | GW | 87.91 | 13.64 | | 76.25 | <50 | <0.5 | <0.5 | 0.6 | -20 | | | |
| | 12/27/1995 | GW | 87.91 | 13.11 | | 74.80 | 6 100 | <25 | <25 | <25 | <50 | 20.000 | | |
| | 6/13/1996 | GW | 87.91 | 10.86 | | 77.05 | 8.300 | <2.5 | <2.5 | <2.5 | <2.5 | 13.000 | | |
| | 12/4/1996 | GW | 87.91 | 13.03 | | 74.88 | 5,900 | <2.5 | <5 | <5 | <5 | 11,000 | | |
| | 6/10/1997 | GW | 87.91 | 10.04 | | 77.87 | <50 | <0.5 | <1 | <1 | <1 | <10 | | |
| | 12/12/1997 | GW | 87.91 | 12.44 | | 75.47 | <50 | <0.5 | <1 | <1 | <1 | <10 | | |
| | 6/18/1998 | GW | 87.91 | 8.89 | | 79.02 | 50 | <0.5 | <1 | <1 | <1 | <10 | | |
| | 3/9/1999 | GW | 87.91 | 10.20 | | 77.71 | 15,000 | <5 | <5 | <5 | <5 | 23,000 | | |
| | 9/28/1999 | GW | 87.91 | 11.81 | | 76.10 | 36,000 | <5 | 12 | 7 | 26 | 35,000 | | 100 |
| | 10/14/1999 | GW | 87.91 | 10.27 | | 77.64 | 1 200 | -0 F | -0 F | 0.51 | -0 F | E 900 | | 100 |
| | 3/21/2000 | GW | 07.91 87.01 | 9.90 | | 76.51 | 1,300 | <0.5 | <0.5 | 0.51 | <0.5 | 5,800 | | |
| | 3/8/2001 | GW | 87.91 | 11.40 | | 76.75 | 20,000 | <0.5 | <0.5 | <0.54 | <0.5 | 29 100 | | |
| | 9/21/2001 | GW | 87.91 | 11.65 | | 76.26 | 5,000 | <0.5 | <0.5 | <0.5 | <1.5 | 6,110 | | |
| | 2/28/2002 | GW | 87.91 | 9.86 | | 78.05 | 3,200 | 35.1 | <0.5 | <0.5 | <1 | 4,620 | | |
| | 9/6/2002 | GW | 87.91 | 12.32 | | 75.59 | 1,900 | <10 | <10 | <10 | <10 | 15,000 | | |
| | 2/19/2003 | GW | 87.91 | 11.63 | | 76.28 | 45,000 | <250 | <250 | <250 | <250 | 32,000 | | |
| | 7/14/2003 | GW | 87.91 | 12.07 | | 75.84 | 9,300 | <500 | <500 | <500 | <500 | 24,000 | <20,000 | |
| | 1/14/2004 | GW | 87.91 | 11.45 | | 76.46 | <50,000 | <500 | <500 | <500 | <500 | 21,000 | <20,000 | |
| | 4/23/2004 | GW | 87.91 | 11.45 | | 76.46 | 5,100 | <250 | <250 | <250 | <250 | 22,000 | 11,000 | |
| | 10/28/2004 | GW | 87.91 | 13.02 | | 75.59 | <5,000 8,500 | <50 | <50 | <50 | <50 | 6,200 | 2,900 | |
| | 1/10/2005 | GW | 87.91 | 14.38 | | 73.53 | <25.000 | <250 | <250 | <250 | <250 | 7,100 | <10.000 | |
| | 4/13/2005 | GW | 87.91 | 14.03 | | 73.88 | <5,000 | <50 | <50 | <50 | <50 | 5,300 | 5,300 | |
| | 7/11/2005 | GW | 87.91 | 11.25 | | 76.66 | <5,000 | <50 | <50 | <50 | <50 | 5,300 | 9,000 | |
| | 10/17/2005 | GW | 87.91 | 12.48 | | 75.43 | <5,000 | <50 | <50 | <50 | <50 | 2,500 | 5,200 | |
| | 1/17/2006 | GW | 87.91 | 10.70 | | 77.21 | <5,000 | <50 | <50 | <50 | <50 | 2,200 | 8,400 | |
| | 7/26/2006 | GW | 87.91 | 10.47 | | 77.44 | 2,700 | <50 | <50 | <50 | <50 | 2,900 | 4,500 | |
| | 10/31/2006 | GW | 87.91 | 12.02 | | 75.89 | 2,300 | <25 | <25 | <25 | <25 | 2,300 | 9,300 | |
| | 1/8/2007 | GW | 87.91 | 11.68 | | 76.23 | 1,500 | <12 | <12 | <12 | <12 | 1,700 | 7,700 | |
| | 7/10/2007 | GW | 87.91 | 11.45 | | 75.40 | 2,300 | <25 | <25 | <25 | <25 | 2 600 | 8,400 | 120 |
| | 10/24/2007 | GW | 87.91 | 12.91 | | 75.00 | 2,800 | <25 | <25 | <25 | <25 | 2,800 | 9,500 | 120 |
| | 1/22/2008 | GW | 87.91 | 12.00 | | 75.91 | <2,500 | <25 | <25 | <25 | <25 | 1,400 | 6,000 | |
| | 4/15/2008 | GW | 87.91 | 11.77 | | 76.14 | 73 | <2.5 | <2.5 | <2.5 | <2.5 | 2,400 | 6,800 | |
| | 7/8/2008 | GW | 87.91 | 12.65 | | 75.26 | 93 | <50 | <50 | <50 | <50 | 2,800 | 7,600 | |
| | 11/19/2008 | GW | 87.91 | 13.98 | | 73.93 | 130 | <50 | <50 | <50 | <50 | 1,900 | 7,100 | |
| | 2/10/2009 | GW | 87.91 | 13.64 | | 74.27 | <50 | <50 | <50 | <50 | <50 | 940 | 2,700 | |
| | 5/7/2009 | GW | 87.91 | 12.00 | | 75.91 | 350 | <20 | <20 | <20 | <20 | 1,900 | 3,900 | |
| | 9/3/2009 | GW | 87.91 | 13.68 | | 74.23 | 890 | <40 | <40 | <40 | <40 | 1,300 | 7,500 | |
| | 2/26/2010 | GW | 07.91 87.01 | 13.00 | | 74.03 | 530 1 100 | <0.50 | <0.50 | <0.50 | <1.0 | 1 100 | 3,900 | |
| | 8/16/2010 | GW | 87.91 | 12.82 | | 75.09 | 1.000 | <10 | <10 | <10 | <20 | 1.100 | 4.800 | |
| | 11/12/2010 | | 87.91 | 12.98 | | 74.93 | .,500 | | | | | ., | .,000 | |
| | 2/3/2011 | GW | 87.91 | 12.38 | | 75.53 | <1000 | <10 | <10 | <10 | <20 | 860 | 3,200 | |
| | 6/23/2011 | | 87.91 | 11.37 | | 76.54 | | | | | | | | |
| | 8/22/2011 | GW | 87.91 | 12.29 | | 75.62 | <250 | <2.5 | <2.5 | <2.5 | <5.0 | 170 | 3,100 | |
| | 2/20/2012 | GW | 87.91 | 13.09 | | 74.82 | <250 | <2.5 | <2.5 | <2.5 | <5.0 | 300 | 2,600 | |
| MW-3 | 11/4/1989 | GW | 87.02 | 15.4 | | /1.62 | <500 | < 0.3 | < 0.3 | <0.3 | < 0.3 | | | |
| | 4/3/1990 | GW | 87.02 | 13.9 | | 73.12 | <100 | <0.5 | <0.5 | <0.5 | <0.5 | | | |
| | 11/20/1990 | GW | 87.02 | 14 67 | | 72.35 | <50 | <0.5 0.3 | <0.5 0.8 | <0.5 | <0.5 1.5 | | | |
| | 3/1/1991 | GW | 87.02 | 15.22 | | 71.80 | <100 | 0.4 | <0.3 | <0.3 | <0.3 | | | |
| 1 | 8/19/1991 | GW | 87.02 | 13.15 | | 73.87 | <30 | <0.3 | <0.3 | <0.3 | <0.3 | | | |

| Location | Sample Date | Notes | TOC Elevation | DTW (ft btoc) | Product Thickness | Water Level Elevation | TPHg | Benzene | Toluene | Ethylbenzen e | Xylene | MTBE | ТВА | TPHd |
|----------|----------------|-------|------------------|------------------|----------------------|-----------------------------|------------|---------|---------|------------------|--------|-------|--------|------|
| | 2410 | | (ft) | (| (ft) | (ft) | | | | hð | ı/L | | | |
| | 11/13/1991 | GW | 87.02 | 15.66 | | 71.36 | <30 | <0.3 | <0.3 | <0.3 | <0.3 | | | |
| | 2/24/1992 | GW | 87.02 | 15.01 | | 72.01 | <50 | 0.65 | 1.4 | 0.66 | 4.4 | | | |
| | 5/19/1992 | GW | 87.02 | 15.52 | | 71.50 | <50 | <0.5 | <0.5 | <0.5 | < 0.5 | | | |
| | 7/22/1992 | GW | 87.02 | 15.63 | | 71.39 | <50 | <0.5 | <0.5 | <0.5 | <0.5 | | | <50 |
| | 6/7/1003 | GW | 07.02 87.02 | 12.13 | | 72.89 | <50 | <0.5 | <0.7 | <0.5 | -0.5 | | | |
| | 12/2/1993 | GW | 87.02 | 13.29 | | 73.73 | <50 | <0.5 | <0.5 | <0.5 | <0.5 | | | |
| | 6/22/1994 | GW | 87.02 | 12.78 | | 74.24 | <50 | <0.5 | <0.5 | <0.5 | <0.5 | | | |
| | 1/10/1995 | GW | 87.02 | 12.01 | | 75.01 | <50 | <0.5 | < 0.5 | <0.5 | <1 | | | |
| | 6/21/1995 | GW | 87.02 | 11.57 | | 75.45 | <50 | <0.5 | <0.5 | <0.5 | <1 | | | |
| | 12/27/1995 | GW | 87.02 | 13.47 | | 73.55 | <50 | <0.5 | <0.5 | <0.5 | <1 | 5.7 | | |
| | 6/13/1996 | GW | 87.02 | 11.22 | | 75.80 | 60 | <0.5 | <0.5 | <0.5 | <0.5 | <10 | | |
| | 12/4/1996 | GW | 87.02 | 13.28 | | 73.74 | <50 | <0.5 | <1 | <1 | <1 | <10 | | |
| | 6/10/1997 | GW | 87.02 | 10.22 | | 76.80 | <50 | <0.5 | <1 | <1 | <1 | <10 | | |
| | 6/18/1008 | GW | 07.02 87.02 | 9.07 | | 77.95 | <50 50 | <0.5 | <1 | <1 | <1 | <10 | | |
| | 3/27/2000 | GW | 87.02 | 13.77 | | 73.25 | <50 | <0.5 | <0.5 | <0.5 | <0.5 | 16 | | |
| | 9/28/2000 | GW | 87.02 | 11.28 | | 75.74 | <50 | <0.5 | 7.4 | <0.5 | 1.3 | 2 | | |
| | 3/8/2001 | GW | 87.02 | 11.75 | | 75.27 | <50 | <0.5 | <0.5 | <0.5 | <0.5 | 60.4 | | |
| | 9/21/2001 | GW | 87.02 | 11.33 | | 75.69 | <50 | <0.5 | <0.5 | <0.5 | <1.5 | 8.18 | | |
| | 2/28/2002 | GW | 87.02 | 10.86 | | 76.16 | <50 | <0.5 | <0.5 | <0.5 | <1 | 25.5 | | |
| | 9/6/2002 | GW | 87.02 | 12.73 | | 74.29 | <50 | 1.2 | <0.5 | <0.5 | 1 | 16 | | |
| | 2/19/2003 | GW | 87.02 | 11.72 | | 75.30 | <500 | <5 | <5 | <5 | <5 | 110 | 00 | |
| | 7/14/2003 | GW | 87.02 | 13.76 | | 73.26 | <50 550 | <0.5 | <0.5 | <0.5 | 0.67 | 28 | <20 | |
| | 1/14/2004 | GW | 07.02 87.02 | 13.17 | | 73.85 | -200 | <0 | <0 | <5 | <0 | 560 | <200 | |
| | 7/1/2004 | GW | 87.02 | 15.19 | | 71.83 | <50 | <0.5 | <0.5 | <0.5 | 0.5 | 48 | <20 | |
| | 10/28/2004 | GW | 87.02 | 15.50 | | 71.52 | <500 | <5 | <5 | <5 | <5 | 290 | <200 | |
| | 1/10/2005 | GW | 87.02 | 15.00 | | 72.02 | <50 | <0.5 | <0.5 | <0.5 | <0.5 | 18 | <20 | |
| | 4/13/2005 | GW | 87.02 | 14.34 | | 72.68 | <50 | <0.5 | <0.5 | <0.5 | <0.5 | 9 | <20 | |
| | 7/11/2005 | GW | 87.02 | 10.82 | | 76.20 | 130 | <1 | <1 | <1 | <1 | 120 | <40 | |
| | 10/17/2005 | GW | 87.02 | 11.84 | | 75.18 | <250 | <2.5 | <2.5 | <2.5 | <2.5 | 260 | <100 | |
| | 1/17/2006 | GW | 87.02 | 11.59 | | 75.43 | 800 | <5 | <5 | <5 | <5 | 980 | 200 | |
| | 4/21/2006 | GW | 87.02 | 10.00 | | 76.22 | <500 | <5 | <5 | <5 | <5 | 48 | <200 | |
| | 7/17/2006 | GW | 87.02 | 9.67 | | 77.35 | 910 810 | <5 | <5 | <5 | <5 | 1,400 | <200 | |
| | 10/31/2006 | GW | 87.02 | 10.85 | | 76.17 | 1.600 | <10 | <10 | <10 | <10 | 2,300 | <400 | |
| | 1/8/2007 | GW | 87.02 | 12.73 | | 74.29 | 520 | <5 | <5 | <5 | <5 | 760 | <200 | |
| | 4/10/2007 | GW | 87.02 | 11.93 | | 75.09 | 630 | <5 | <5 | <5 | <5 | 750 | <200 | |
| | 7/10/2007 | GW | 87.02 | 11.30 | | 75.72 | 1,800 | <5 | <5 | <5 | <5 | 2,400 | <200 | 66 |
| | 10/24/2007 | GW | 87.02 | 13.77 | | 73.25 | 2,000 | <25 | <25 | <25 | <25 | 3,500 | <1,000 | |
| | 1/22/2008 | GW | 87.02 | 12.92 | | 74.10 | 1,600 | <12 | <12 | <12 | <12 | 2,800 | <500 | |
| | 4/15/2008 | GW | 87.02 | 15.25 | | 71.77 | <50 | <2.5 | <2.5 | <2.5 | <2.5 | 960 | <50 | |
| | 11/10/2008 | GW | 87.02 | 12.27 | | 74.75 | <50 | <50 | <50 | <50 | <50 | 2,200 | <1,000 | |
| | 2/10/2009 | GW | 87.02 | 13.61 | | 73.41 | <50 | <50 | <50 | <50 | <50 | 2,700 | <1,000 | |
| | 5/7/2009 | GW | 87.02 | 11.75 | | 75.27 | 140 | <10 | <10 | <10 | <10 | 780 | <200 | |
| | 9/3/2009 | GW | 87.02 | 13.47 | | 73.55 | 1,100 | <10 | <10 | <10 | <10 | 2,400 | <200 | |
| | 10/29/2009 | GW | 87.02 | 13.04 | | 73.98 | 1,000 | <10 | <10 | <10 | <20 | 1,500 | 110 | |
| | 2/26/2010 | GW | 87.02 | 12.44 | | 74.58 | 1,500 | <10 | <10 | <10 | <20 | 1,500 | <80 | |
| | 8/16/2010 | GW | 87.02 | 11.43 | | 75.59 | 1,900 | <0.50 | <0.50 | <0.50 | <1.0 | 2,400 | 20 | |
| | 11/12/2010 | 0.04 | 87.02 | 12.05 | | 74.97 | 4000 | 10 | 10 | 10 | | 4 500 | 450 | |
| | 2/3/2011 | GW | 87.02 | 12.31 | | 74.71 | <1000 | <10 | <10 | <10 | <20 | 1,500 | 150 | |
| | 8/22/2011 | GW | 87.02 | 11.54 | | 75.48 | <1000 | <10 | <10 | <10 | <20 | 2 600 | <80 | |
| | 2/20/2012 | GW | 87.02 | 12.81 | | 74.21 | <1000 | <10 | <10 | <10 | <20 | 1.700 | 110 | |
| MW-4 | 11/12/2010 | GW | 78.06 | | | | <50 | < 0.50 | < 0.50 | < 0.50 | <1.0 | 95 | 6.9 | |
| | 2/3/2011 | GW | 78.06 | 12.09 | | 65.97 | <50 | <0.50 | <0.50 | <0.50 | <1.0 | 110 | 12 | |
| | 6/23/2011 | GW | 78.06 | 11.33 | | 66.73 | <50 | <0.50 | <0.50 | <0.50 | <1.0 | 36 | <4.0 | |
| | 8/22/2011 | GW | 78.06 | 12.09 | | 65.97 | <50 | < 0.50 | < 0.50 | <0.50 | <1.0 | 3.7 | <4.0 | |
| 00.3 | 2/20/2012 | GVV | 78.06 | 12.61 | | 00.40 | <50 | <0.50 | <0.50 | <0.50 | <1.0 | 12 | <4.0 | |
| QU-2 | 6/7/1003 | | | | | | <50 | <0.5 | <0.5 | <0.5 | <0.5 | | | |
| | 12/2/1993 | | | | | | <50 | <0.5 | <0.5 | <0.5 | <0.5 | | | |
| | 6/22/1994 | | | | | | <50 | <0.5 | <0.5 | <0.5 | <0.5 | | | |
| | 1/10/1995 | | | | | | <50 | <0.5 | <0.5 | <0.5 | <1 | | | |
| | 6/21/1995 | | | | | | <50 | <0.5 | <0.5 | <0.5 | <1 | | | |
| | 12/27/1995 | | | | | | <50 | <0.5 | <0.5 | <0.5 | <1 | <5 | | |
| 1 | 6/13/1996 | 1 | 1 | l I | 1 | | <50 | < 0.5 | < 0.5 | <0.5 | < 0.5 | <10 | | |

Notes: P = well purged prior to sampling NP = well not purged prior to sampling TOC = Top of Casing

| Location | Sample Date | Notes | TOC Elevation | DTW (ft btoc) | Product Thickness | Water Level Elevation | TPHg | Benzene | Toluene | Ethylbenzen e | Xylene | МТВЕ | ТВА | TPHd |
|----------|----------------|-------|------------------|------------------|----------------------|-----------------------------|------|---------|---------|------------------|--------|------|-----|------|
| | | | (11) | | (11) | (ft) | | μg/L | | | | | | |

DTW = Depth to Water ft = feet

tt = feet TPHg = Total Petroleum Hydrocarbons as Gasoline MTBE = Methyl tert-butyl ether TPHd = Total Petroleum Hydrocarbons as Diesel µg/L = micrograms per liter - = not analyzed < = analyte not detected, result is less than value provided

Table 3: Historical Groundwater Flow Directions and Gradients Former BP Service Station No. 11102 100 MacArthur Blvd, Oakland, CA Local Case #RO456

| | | Approximate | | | | |
|--------------|--------------------------|------------------|--|--|--|--|
| Date Sampled | Approximate Flow | Hydraulic | | | | |
| • | Direction | Gradient (ft/ft) | | | | |
| 11/11/1989 | Southwest | 0.04 | | | | |
| 4/3/1990 | Southwest | 0.07 | | | | |
| 7/30/1990 | Southwest | 0.07 | | | | |
| 11/20/1990 | Southwest | 0.09 | | | | |
| 8/19/1991 | South | 0.06 | | | | |
| 11/13/1991 | South | 0.06 | | | | |
| 2/2//1002 | Southwest | 0.00 | | | | |
| 5/10/1002 | Southwest | 0.00 | | | | |
| 7/22/1002 | West-Southwest | 0.00 | | | | |
| 8/1//1002 | West-Southwest | 0.07 | | | | |
| 11/11/1002 | Southwest | 0.07 | | | | |
| 6/7/1003 | Southwest | 0.00 | | | | |
| 12/2/1002 | Southwest | 0.07 | | | | |
| 6/22/1993 | Southwest | 0.05 | | | | |
| 0/22/1994 | Southwost | 0.05 | | | | |
| 6/21/1005 | West Southwest | 0.05 | | | | |
| 0/21/1995 | West-Southwest | 0.06 | | | | |
| 6/12/1006 | West-Southwest | 0.06 | | | | |
| 0/13/1990 | West-Southwest | 0.06 | | | | |
| 12/4/1990 | West-Southwest | 0.05 | | | | |
| 6/10/1997 | West-Southwest | 0.05 | | | | |
| 12/12/1997 | West-Southwest | 0.05 | | | | |
| 6/18/1998 | West-Southwest | 0.05 | | | | |
| 3/9/1999 | West-Southwest | 0.07 | | | | |
| 9/28/1999 | West-Southwest | 0.07 | | | | |
| 3/27/2000 | West-Southwest | 0.08 | | | | |
| 9/28/2000 | North-Northwest | 0.03 | | | | |
| 3/8/2001 | North-Northwest | 0.04 | | | | |
| 9/21/2001 | VVest North Northwest | 0.03 | | | | |
| 2/28/2002 | North-Northwest | 0.04 | | | | |
| 2/19/2003 | West | 0.04 | | | | |
| 1/14/2003 | West | 0.06 | | | | |
| 1/14/2004 | West-Nort | 0.06 | | | | |
| 4/23/2004 | West | 0.05 | | | | |
| 10/28/2004 | West | 0.00 | | | | |
| 1/10/20/2004 | West-Northwest | 0.07 | | | | |
| 1/10/2005 | West Southwest | 0.07 | | | | |
| 7/11/2005 | West-Southwest | 0.00 | | | | |
| 10/17/2005 | Southwest | 0.00 | | | | |
| 1/17/2005 | West | 0.03 | | | | |
| 7/26/2006 | Southwest | 0.04 | | | | |
| 10/31/2006 | Southwest | 0.03 | | | | |
| 1/8/2007 | West | 0.04 | | | | |
| 1/0/2007 | West | 0.00 | | | | |
| 7/10/2007 | Southwest | 0.03 | | | | |
| 10/24/2007 | West-Southwest | 0.04 | | | | |
| 1/22/2008 | West-Oouthwest | 0.00 | | | | |
| 1/22/2000 | West-Southwest | 0.00 | | | | |
| 7/8/2008 | West-Southwest | 0.05 | | | | |
| 11/10/2008 | West | 0.05 | | | | |
| 2/10/2009 | West | 0.00 | | | | |
| 5/7/2009 | West | 0.05 | | | | |
| 9/3/2009 | West | 0.05 | | | | |
| 10/29/2009 | West | 0.00 | | | | |
| 2/26/2010 | West | 0.05 | | | | |
| 8/16/2010 | West-Southwest | 0.05 | | | | |
| 2/3/2011 | West-Southwest | 0.00 | | | | |
| 6/23/2011 | West-Southwest | 0.05 | | | | |
| 8/22/2011 | West-Southwest | 0.05 | | | | |
| 2/20/2012 | West-Southwest | 0.05 | | | | |

Table 4: Most Recent Maximum Concentration of Contaminants Detected in Soil and Groundwater Former BP Service Station No. 11102 100 MacArthur Blvd, Oakland, CA Local Case #R0456

| | | | | Soil ¹ | | Groundwater | | | | | | | |
|--------------|---|-------------------------------|----------------|---|-------------------------------|----------------|------------------------------|---|----------------|--|----------------|---|---|
| Analyte | Most Recent Concentration Observed (mg/kg) | Sample Depth (feet bgs) | Sample Date | Maximum Concentration Observed (mg/kg) | Sample Depth (feet bgs) | Sample Date | Commercial ESL (mg/kg) | Most Recent Concentratio n Observed (µg/L) | Sample Date | Maximum Concentration Observed (µg/L) | Sample Date | Commercial ESL for the Protection of the Vapor Intrusion Pathway (µg/L) | State of California Maximum Contaminant Level (µg/L) |
| TPHg | <0.3 (MW-4) | 6.5 | 10/6/2010 | 1,300 (SB-7) | 2.0 | 7/14/2005 | 450 | 1,900 (MW-3) | 8/16/2010 | 45,000 (MW-2) | 2/19/2003 | 29 | NA |
| Benzene | <0.0050 (MW-4) | 6.5 | 10/6/2010 | 0.08 (MW-2) | 10.0 | 10/25/1989 | 0.27 | 1.6 (MW-1) | 5/7/2009 | 440 (MW-1) | 5/19/1992 | 1,800 | 1 |
| Toluene | <0.0050 (MW-4) | 6.5 | 10/6/2010 | 0.006 (MW-3) | 5.0 | 10/26/1989 | 210 | 1.0 (MW-1) | 4/21/2006 | 110 (MW-1) | 2/19/2003 | 530,000 | 150 |
| Ethylbenzene | <0.0050 (MW-4) | 6.5 | 10/6/2010 | 3.0 (SB-7) | 2.0 | 7/14/2005 | 5 | 0.83 (MW-1) | 1/22/2008 | 250 (MW-1) | 5/19/1992 | 170,000 | 300 |
| Xylenes | <0.01 (MW-4) | 6.5 | 10/6/2010 | 3.9 (SB-7) | 5.0 | 7/14/2005 | 100 | 0.89 (MW-1) | 9/3/2009 | 1,100 (MW-1) | 3/9/1999 | 160,000 | 1750 |
| MTBE | <0.0050 (MW-4) | 6.5 | 10/6/2010 | 0.37 (SB-4) | 9.5 | 7/14/2005 | 65 | 1,700 (MW-3) | 2/20/2012 | 49,000 (MW-1) | 3/9/1999 | 80,000 | 13 |
| TPHd | 2,100 (TD-1) | 0.5 | 11/22/1994 | 2,100 (TD-1) | 0.5 | 11/22/1994 | 450 | 160 (MW-1) | 7/10/2007 | 2900 (MW-1) | 6/18/1998 | NA | NA |

¹ Soil results are reported from the vadose zone, which does not exceed 10 feet bgs

² ESL Table K-2 - Direct Exposure Soil Screening Levels Commercial/Industrial Exposure

³ ESL Table E-1 - Groundwater Screening Levels for Evaluation of Potential Vapor Intrusion Concerns

Bold = exceedances of commercial ESL

TPHg = Total Petroleum Hydrocarbons as Gasoline

MTBE = methyl tert-butyl ether

TPHd = Total Petroleum Hydrocarbons as Diesel

mg/kg = milligrams per kilogram

 μ g/L = micrograms per liter

bgs = below ground surface

ND = non-detect, below laboratory detection limits

GRO Commercial ESL values are listed as TPH (gasolines) in Table A of Screening for Environmental Concerns at Sites with Contaminated Soil and Groundwater

DRO Commercial ESL values are listed as TPH (middle distillates) in Table A of Screening for Environmental Concerns at Sites with Contaminated Soil and Groundwater

NA = not available

Figures


PLOTTED: 3/8/2012 2:32 PM BY: BEARDSLEY, DANIEL PLOTSTYLETABLE: ARCADIS-EMV.CTB PAGESETUP: LYR:(Opt)ON=*;0FF=*REF* D: 3/8/2012 2:31 PM ACADVER: 18.1S (LMS TECH) SAVED: PIC: S.GLENN PM: H.PHILLIPS TM: C.MEYER rt(GP09BPNAC11-B-01.dwg LAYOUT: 1 SAVE) LD:--DB: J. HARRIS 11\B0000\Closuri CITY: PETALUMA, CA DIV/GROUP: ENV G:\ENVCAD\Emeryville\ACT\GP09BPNA\C1



- - PROPERTY BOUNDARY

⊕ BP MONITORING WELL LOCATION

BP SOIL BORING LOCATION

⊕ UNOCAL MONITORING WELL LOCATION

▲ UNOCAL SOIL BORING LOCATION

NOTES:

- BASE MAP PREPARED BY DIGITIZING A HARD COPY OF A DRAWING BY "BROADBENT AND ASSOCIATES, INC"., TITLED "SITE LAYOUT PLAN WITH PROPOSED SOIL BORING AND WELL LOCATIONS", DATED 3/9/09, AT A SCALE OF 1'=40'
- 2. ALL LOCATIONS ARE APPROXIMATE.
- 3. THE NEW MONITORING WELL WAS INSTALLED ON 10/06/2010 AND DEVELOPED AND SAMPLED ON 10/12/2010.



FORMER BP SERVICE STATION #11102 100 MACARTHUR BOULEVARD, OAKLAND, CALIFORNIA **RECOMMENDATION FOR CASE CLOSURE**

SITE MAP WITH SOIL BORING AND MONITORING WELL LOCATIONS





| | PROPERTY BOUNDARY |
|----------|---|
| \oplus | BP MONITORING WELL LOCATION |
| A | BP SOIL BORING LOCATION |
| \oplus | UNOCAL MONITORING WELL LOCATION |
| ▲ | UNOCAL SOIL BORING LOCATION |
| 100 | TPHg ISOCONCENTRATION CONTOUR (mg/Kg) (DASHED WHERE INFERRED) |
| (<1.0) | TPHg CONCENTRATION IN mg/Kg in 1989 |
| (<1.0) | TPHg CONCENTRATION IN mg/Kg in 1994 |
| (<1.0) | TPHg CONCENTRATION IN mg/Kg in 2005 |
| (<1.0) | TPHg CONCENTRATION IN mg/Kg in 2010 |
| TPHg | TPHg TOTAL PETROLEUM HYDROCARBONS AS GASOLINE |
| (mg/K | g) MILLIGRAMS PER KILOGRAM |
| (NS) | NOT SAMPLED AT THE SPECIFIC INTERVAL |
| | |

NOTES:

- 1. BASE MAP PREPARED BY DIGITIZING A HARD COPY OF A DRAWING BY "BROADBENT AND ASSOCIATES, INC"., TITLED "SITE LAYOUT PLAN WITH PROPOSED SOIL BORING AND WELL LOCATIONS", DATED 3/9/09, AT A SCALE OF 1'=40'
- 2. ALL LOCATIONS ARE APPROXIMATE.



FORMER BP SERVICE STATION #11102 100 MACARTHUR BOULEVARD, OAKLAND, CALIFORNIA **RECOMMENDATION FOR CASE CLOSURE**

HISTORICAL LATERAL EXTENT OF TPHg SOIL IMPACTS





| | | PROPERTY BOUNDARY |
|-----|----------|--|
| | ⊕ | BP MONITORING WELL LOCATION |
| | | BP SOIL BORING LOCATION |
| | \oplus | UNOCAL MONITORING WELL LOCATION |
| | | UNOCAL SOIL BORING LOCATION |
| 100 | | TPHg ISOCONCENTRATION CONTOUR (mg/Kg) (DASHED WHERE INFERRED) |
| | (<1.0) | MTBE CONCENTRATION IN mg/Kg in 1989 |
| | (<1.0) | MTBE CONCENTRATION IN mg/Kg in 1994 |
| | (<1.0) | MTBE CONCENTRATION IN mg/Kg in 2005 |
| | (<1.0) | MTBE CONCENTRATION IN mg/Kg in 2010 |
| | MTBE | METHYL TERTIARY-BUTYL ETHER |
| | (mg/Kg) | MILLIGRAMS PER KILOGRAM |
| | (NS) | NOT SAMPLED AT THE SPECIFIC INTERVAL |
| | | |

NOTES:

- 1. BASE MAP PREPARED BY DIGITIZING A HARD COPY OF A DRAWING BY "BROADBENT AND ASSOCIATES, INC"., TITLED "SITE LAYOUT PLAN WITH PROPOSED SOIL BORING AND WELL LOCATIONS", DATED 3/9/09, AT A SCALE OF 1'=40'
- 2. ALL LOCATIONS ARE APPROXIMATE.



FORMER BP SERVICE STATION #11102 100 MACARTHUR BOULEVARD, OAKLAND, CALIFORNIA **RECOMMENDATION FOR CASE CLOSURE**

HISTORICAL LATERAL EXTENT OF MTBE SOIL IMPACTS





| | LEGEND: |
|----------|---------------------------------|
| | PROPERTY BOUNDARY |
| \oplus | BP MONITORING WELL LOCATION |
| | BP SOIL BORING LOCATION |
| \oplus | UNOCAL MONITORING WELL LOCATION |
| A | UNOCAL SOIL BORING LOCATION |
| A A' | CROSS SECTION LOCATION |

- 1. BASE MAP PREPARED BY DIGITIZING A HARD COPY OF A DRAWING BY "BROADBENT AND ASSOCIATES, INC"., TITLED "SITE LAYOUT PLAN WITH PROPOSED SOIL BORING AND WELL LOCATIONS", DATED 3/9/09, AT A SCALE OF 1'=40'
- 2. ALL LOCATIONS ARE APPROXIMATE.









OFF Opt)ON=*; LYR M PM:(Reqd) TM:(Opt) PIC:(Opt)





 BASE MAP PREPARED BY DIGITIZING A HARD COPY OF A DRAWING BY "BROADBENT AND ASSOCIATES, INC"., TITLED "SITE LAYOUT PLAN WITH PROPOSED SOIL BORING AND WELL LOCATIONS", DATED 3/9/09, AT A SCALE OF 1'=40'



FORMER BP SERVICE STATION #11102 100 MACARTHUR BOULEVARD, OAKLAND, CALIFORNIA **RECOMMENDATION FOR CASE CLOSURE**



ARCADIS

FIGURE





 BASE MAP PREPARED BY DIGITIZING A HARD COPY OF A DRAWING BY "BROADBENT AND ASSOCIATES, INC"., TITLED "SITE LAYOUT PLAN WITH PROPOSED SOIL BORING AND WELL LOCATIONS", DATED 3/9/09, AT A SCALE OF 1'=40'



FORMER BP SERVICE STATION #11102 100 MACARTHUR BOULEVARD, OAKLAND, CALIFORNIA **RECOMMENDATION FOR CASE CLOSURE**

EXTENT OF MTBE GROUNDWATER IMPACTS (FEBRUARY 2012)



PROPERTY BOUNDARY
 ● BP MONITORING WELL LOCATION

 (<50) TPHg CONCENTRATION IN MICROGRAMS PER LITER (µg/L)
 TPHg TOTAL PETROLEUM HYDROCARBONS AS GASOLINE
 APPROXIMATE GROUNDWATER FLOW DIRECTION

NOTES:

 BASE MAP PREPARED BY DIGITIZING A HARD COPY OF A DRAWING BY "BROADBENT AND ASSOCIATES, INC"., TITLED "SITE LAYOUT PLAN WITH PROPOSED SOIL BORING AND WELL LOCATIONS", DATED 3/9/09, AT A SCALE OF 1'=40'



FORMER BP SERVICE STATION #11102 100 MACARTHUR BOULEVARD, OAKLAND, CALIFORNIA **RECOMMENDATION FOR CASE CLOSURE**

EXTENT OF TPHg GROUNDWATER IMPACTS (AUGUST 2011)

(AUGUST 2011)





1. BASE MAP PREPARED BY DIGITIZING A HARD COPY OF A DRAWING BY "BROADBENT AND ASSOCIATES, INC"., TITLED "SITE LAYOUT PLAN WITH PROPOSED SOIL BORING AND WELL LOCATIONS", DATED 3/9/09, AT A SCALE OF 1'=40'



FORMER BP SERVICE STATION #11102 100 MACARTHUR BOULEVARD, OAKLAND, CALIFORNIA **RECOMMENDATION FOR CASE CLOSURE**

EXTENT OF MTBE GROUNDWATER IMPACTS (AUGUST 2011)

ARCADIS



| | LEGEND: |
|-------|---|
| | PROPERTY BOUNDARY |
| Φ | BP MONITORING WELL LOCATION |
| (<50) | TPHg CONCENTRATION IN MICROGRAMS PER LITER ($\mu\text{g/L})$ |
| TPHg | TOTAL PETROLEUM HYDROCARBONS AS GASOLINE |
| | APPROXIMATE GROUNDWATER FLOW DIRECTION |

 BASE MAP PREPARED BY DIGITIZING A HARD COPY OF A DRAWING BY "BROADBENT AND ASSOCIATES, INC"., TITLED "SITE LAYOUT PLAN WITH PROPOSED SOIL BORING AND WELL LOCATIONS", DATED 3/9/09, AT A SCALE OF 1'=40'



FORMER BP SERVICE STATION #11102 100 MACARTHUR BOULEVARD, OAKLAND, CALIFORNIA **RECOMMENDATION FOR CASE CLOSURE**

EXTENT OF TPHg GROUNDWATER IMPACTS (FEBRUARY 2011)

ARCADIS

FIGURE



| LEGEND: |
|--|
| PROPERTY BOUNDARY |
| BP MONITORING WELL LOCATION |
| MTBE CONCENTRATION IN MICROGRAMS PER LITER ($\mu g/L)$ |
| METHYL TERTIARY-BUTYL ETHER |
| MTBE ISOCONCENTRATION CONTOUR (µg/L) (DASHED WHERE INFERRED) |
| APPROXIMATE GROUNDWATER FLOW DIRECTION |
| |

 BASE MAP PREPARED BY DIGITIZING A HARD COPY OF A DRAWING BY "BROADBENT AND ASSOCIATES, INC"., TITLED "SITE LAYOUT PLAN WITH PROPOSED SOIL BORING AND WELL LOCATIONS", DATED 3/9/09, AT A SCALE OF 1'=40'



FORMER BP SERVICE STATION #11102 100 MACARTHUR BOULEVARD, OAKLAND, CALIFORNIA **RECOMMENDATION FOR CASE CLOSURE**

EXTENT OF MTBE GROUNDWATER IMPACTS (FEBRUARY 2011)

ARCADIS

FIGURE







LEGEND:

Ð

PROPERTY BOUNDARY

BP MONITORING WELL LOCATION

 BASE MAP PREPARED BY DIGITIZING A HARD COPY OF A DRAWING BY "BROADBENT AND ASSOCIATES, INC"., TITLED "SITE LAYOUT PLAN WITH PROPOSED SOIL BORING AND WELL LOCATIONS", DATED 3/9/09, AT A SCALE OF 1'=40'



POTENTIOMETRIC SURFACE CONTOURS (FEBRUARY 2012)

ARCADIS

FIGURE



EXPLANATION

- X INCOMPLETE PATHWAY
- O COMPLETE PATHWAY





ARCADIS

Appendix A

Boring Logs

ALTON GEOSCIENCE BORING LOG

PROJECT: 30-063

TYPE 10" HSA

4

BORING DATE: 10-26-89

LOCATION: 100 MacArthur Boulevard, Oakland

GEOLOGIST: M. Hopwood

DRILLING COMPANY: Bay Area Exploration

| | ۱ | BLOW CTS | MATERIAL ENCOUNTERED | uscs |
|--|----------------------------------|---|---|------|
| - | | | Asphalt Over Road Base Loose, dry, tan to orange, gravelly SAND. | GP |
| | | 5,14,16 | Loose, damp, tan to orange, gravelly SAND; wood fragments. CGI = ND. | GP |
| | | 6,10,13 | Loose, damp, tan to light brown, clayey SAND; poorly sorted. OGI = ND. | SC |
| - 15 - | | 8,8,25 | Loose, very moist, tan to light brown, clayey SAND; some iron staining. | SC |
| 20 20 25 25 | | 9,9,12 | Loose, saturated, tan to brown, gravelly SAND, with clay. CGI = ND. | GC |
| - 30 - | | | Medium stiff, moist, tan CLAY. | CL |
| - | | | Total Depth = 32 Feet | |
| TPH =Tol TRPH = T V = Gi ND = Not CGI = Co | Lal Pe Fotal Found Dete | Leven Hydroc Recoverable Pe I Water Piezomu cted stible Gas Indic | tarbons ++ = Sample Analyzed for Hydrocarbon B = Benzene tarbons T = Toluene atroleum Hydrocarbons Concentration T = Toluene atroleum Hydrocarbons L = Sampling Interval E = Ethylbenzene atroleum = Parts per Million X = Xylene ator LEL = Lower Explosive Limit Total Depth = 32 F | eet |

MONITORING WELL CONSTRUCTION DETAIL



ALTON GEOSCIENCE BORING LOG

| PROJECT: | 30-063 | BORING DATE: 10-25-89 | | | | |
|----------|----------------------------------|-----------------------|--|--|--|--|
| | 100 MacArthur Boulevard, Oakland | GEOLOGIST: M. Hopwood | | | | |
| TYPE: | 10" HSA | | | | | |

DRILLING COMPANY: Bay Area Exploration

| DEPTH (FEET) | i | BLOW CTS | MATERIAL ENCOUNTERED | USCS |
|---|--|---|--|------|
| | | 3,4,7 | Asphalt Over Road Base Very loose, damp, dark brown, silty CLAY. Loose, damp, greenish gray, silty CLAY with some coarse sand; very slight odor. CGI = 75 ppm. | CL |
| | T | 2,4,6 | Medium stiff, damp, tan, sandy SILTY/CLAY. CGI = ND. | cr |
| | | 5,7,12 | Moderately stiff, damp, tan, clayey SILT. | ML |
| | | | Stiff, damp, gray, silty CLAY; iron stains; calcite stringers. | CL |
| - - | | | Total Depth - 32 Feet | |
| TPH =Tota TRPH = To V = Gr NO = Not (CGI = Con | I Per stal F sund Detec nbus | 1 troleum Hydroc: Recoverable Pe Water Piezome sted tible Gas Indica | 1 ++ = Sample Analyzod for Hydrocarbon B = Benzene arbons ++ = Sample Analyzod for Hydrocarbon T = Toluene troleum Hydrocarbons Concentration T = Toluene trois Surface I = Sampling Interval E = Ethylbenzene ppm = Parts per Million X = Xylene ttor LEL = Lower Explosive Limit Total Depth = 32 Fe | et |

MONITORING WELL CONSTRUCTION DETAIL



.....

ALTON GEOSCIENCE BORING LOG

PROJECT: 30-063

BORING DATE: 10-26-89

LOCATION: 100 MacArthur Boulevard, Oakland

GEOLOGIST: M. Hopwood

TYPE: 10" HSA

BORING NO. : MM-3

DRILLING COMPANY: Bay Area Explored

| DEPTH (FEET) | ł | BLOW CTS | MATERIAL ENCOUNTERED | uscs |
|---|---|--|--|--------|
| - | | | Asphalt Over Road Base Loose, dry, tan to orange, gravelly SAND. | GP |
| - 5 | | 7,11,14 | Moderately stiff, damp, tan to gray/green, silty CLAY, with gravel; some iron staining. | ar |
| - - - 15 - | | 3,5,6 6,8,13 | Moderately soft, damp, brown, silty CLAY. | CL |
| - - 20 | | | Moderately loose, damp, tan to brown, sandy CLAY. | GT |
| - - 25 | | | Soft, moist, tan CIAY. | |
| - | | | Becomes silty. | |
| - - | | | Total Depth - 32 Feet | |
| - - | | | | |
| TPH =Tota TRPH = Tota ∇ = Group ND = Not f CGI = Con | ai Pe otai I ound Dete nous | troleum Hydroca Recoverable Pa Water Piezome cted tible Gas Indica | arbons ++ = Sample Analyzed for Hydrocarbon B = Benzene troleum Hydrocarbons Concentration T = Toluene tric Surface I = Sampling Interval E = Ethylbenzene ppm = Parts per Million X = Xylene tor LEL = Lower Explosive Limit Total Depth = 32 Fermion | et |

MONITORING WELL CONSTRUCTION DETAIL



--



Project: GP09BPNA.C111.C0000 Template: C:\Documents and Settings\lkwong\Desktop\Boring Logs\BP 11102\boring_well2008 (1).ldfx Data File: MW-4.dat Date: 11/8/2010

| DC | 1333 Broadway, Suite 80 | 00 | LOC | G O | FΒ | ORI | ١G | |
|--|--|--|---|-----------------------|------------|-----------|----------|---|
| URD | Oakland, California 9461 | 2 | Borehole IL Total Depth | 5: <u>58</u> 1: 19 | ft bg: | | | |
| PROJECT INF | FORMATION | DRILL | ING | NFO | RMATI | ON | | |
| Project: Former BP Service | Station #11102 | Drilling | Company: Gre | egg Dr | illing | & Testin | g | |
| Site Location: 100 MacArth | ur Boulevard, Oakland, CA | Driller | : Paul Rogers | | | | | |
| Project Manager: Lynelle (| Dnishi | Туре с | of Drilling Rig: M | IARL | M10T | | | |
| PG: Barbara Jakub | | Drilling | g Method: Airkni | ife, Ha | ind Au | ger (HA |), and] | Hydropunch (HP) |
| Geologist: Jeremy Quick | | Sampl | ing Method: | | | | | |
| JOD NUMBER: 3848/349.0A | | | Drilled: Octobe | er 7, 20 | 005 | | | |
| Groundwater Depth: Grou | Indwater Not Encountered | Boring | Location: MacA | rthur | Blvd | annrox | 175 ft | north of Oakland Ave |
| Air Knife or Hand Auger D | epth: 12 ft bgs (HA) | Boring | Diameter: 3.25 | " (HA |), 2.75 | " (HP) | 170 10 | norm of Ouxiand Arve. |
| Coordinates: X NA | Y NA | Boring | Type: Explorato | ry HP | Borin | g | | |
| (s | | | | | | | | |
| Depth (ft bg Symbol | Lithologic Descriptic | n | | nscs | (mqq) UI q | Sample ID | Recovery | Comments |
| - 4 - 4 - 4 - 6 - 8 SILT: 1 - 10 - 0 CONCF - SAND: - CLAYE - | RETE (FILL) Black (2.5Y 2.5/1), loose, moist, 1 EY SILT: Black (2.5Y 2.5/1), very soft to s 0% silt, 5% sand, medium to high plastic ft bgs: Color change to grayish brown (pased clay (20%), increased silt (75%). ELLY SILT: Grayish brown (2.5Y 5/2), lo 0% silt, 5% sand, 30% gravel. 6.5 ft bgs: ~5-25 mm subangular chert of 7 ft bgs: Color change to light olive browr ased silt (60%), decreased gravel (20%). Light olive brown (2.5Y 5/3) to light yellor medium stiff, damp, 10% clay, 80-85% s , 5-10% sand (increasing with depth), lo 8-12 ft bgs: Color change to grayish bro Medium stiff to stiff. | 10% silt, 9 soft, damp ity. (2.5Y 5/2). ose, soft, clasts thro n (2.5Y 5/3 wish brow silt (decrea w to no pl wish brow silt (decrea w to no pl | 0% sand. o to moist, 25% damp, 15% oughout. 3). m (2.5Y 6/3), asing with asticity. 5/2). No | SP ML ML | | | | Boring grouted with neat Portland Cement. Top 3-6" finished to grade with cement. |

| UR | JRS LOG OF BORING Borehole ID: SB-1 | | | | | | | |
|----------------|--|---|------|-----------|-------------|----------|---------------------------------|--|
| Depth (ft bgs) | Symbol | Lithologic Description | nscs | PID (ppm) | Sample I.D. | Recovery | Comments | |
| - 12 | | HydroPunch driven and exposed from 12 to 14 ft bgs. After 1 hour, no water was available for sampling. | | | | | | |
| - 14 | | HydroPunch driven and exposed from 14 to 16 ft bgs. After 1 hour, no water was available for sampling. | | | | | | |
| - 16 | | HydroPunch driven and exposed from 17 to 19 ft bgs. After 1 hour, no water was available for sampling. | | | | | | |
| | | | | | | | Bottom of Boring = 19 ft bgs | |
| | | | | · | · | | | |
| | | | | | | | | |
| | | | | | | | | |
| | | | | | | | | |
| | | | | | | | | |

Page 2 of 2

Borehole ID: SB-1

BP/Atlantic Richfield Company

| 1333 Broadway, Suite 80 Oakland, California 9461PROJECT INFORMATIONProject:Former BP Service Station #11102Site Location:100 MacArthur Boulevard, Oakland, CAProject Manager:Lynelle OnishiPG:Barbara JakubGeologist:Barbara Jakub / Jeremy QuickJob Number:38487349.0A022BORING INFGroundwater Depth:Groundwater Not EncounteredAir Knife or Hand Auger Depth:8 ft bgs (HA)Coordinates:XNAYNA | | | | LOG OF BORING Borehole ID: SB-2 Total Depth: 19 ft bgs DRILLING INFORMATION DRILLING INFORMATION Drilling Company: Gregg Drilling & Testing Driller: Paul Rogers Type of Drilling Rig: MARL M10T Drilling Method: Airknife, Hand Auger (HA), and Hydropunch (HP) Sampling Method: Date(s) Drilled: October 7, 2005 INFORMATION Boring Location: MacArthur Blvd., approx. 125 ft north of Oakland 4 Boring Diameter: 3.25 " (HA), 2.75 " (HP) | | | | | | Hydropunch (HP) north of Oakland Ave. |
|---|-------|---|---|--|------------|----------|------------------|-----------|----------|---|
| Depth (ft bgs) | 5 | | Lithologic Descriptic |)n | | nscs | PID (mqq) CII | Sample ID | Recovery | Comments |
| | CONCF | ETE ELLY SILT: Fill EY SILT: Black bgs: 15% clay bgs: 15% clay fl bgs: Oxidat fl bgs: Color 0%), decrease m stiff to stiff, r | material. (2.5Y 2.5/1), soft, dry to n y, 83% silt, 2% sand, and ange to dark gray (2.5Y 4/ ion staining is apparent. change to grayish brown (ad silt (80%), no sand or g nedium plasticity. | noist. 3% angul: /1). Minor (2.5Y 5/2) ravel. Dr | ar gravel. | ML ML | 4.6 14.0 0 | | | Boring grouted with neat Portland Cement. Top 3-6" finished to grade with cement. |

Borehole ID: SB-2

| UR | | LOG OF BORING | Borehole ID: SB-2 | | | | | | |
|----------------|--------|---|-------------------|-----------|-------------|----------|--------------------|--|--|
| Depth (ft bgs) | Symbol | Lithologic Description | nscs | PID (ppm) | Sample I.D. | Recovery | Comments | | |
| - 12 | | HydroPunch driven and exposed from 12 to 14 ft bgs. After 1 hour, no water was available for sampling. | | | | - | | | |
| - 14 | | HydroPunch driven and exposed from 14 to 16 ft bgs. After 1 hour, no water was available for sampling. | | | | | | | |
| - 18 | | HydroPunch driven and exposed from 17 to 19 ft bgs. After 1 hour, no water was available for sampling. | | | | | Bottom of Boring = | | |
| | | | | | L | | | | |
| BP/Atlant | tic R | Richfield Company Page 2 of 2 | | Bo | orehole | D: | SB-2 | | |

| LOG OF BORINGBorehole ID: SB-3Oakland, California 94612DROJECT INFORMATIONDRILLING INFORMATIONProject: Former BP Service Station #11102Drilling Company: Gregg Drilling & TestingSite Location: 100 MacArthur Boulevard, Oakland, CADriller: Paul RogersProject Manager: Lynelle OnishiType of Drilling Rig: MARL M10TPG: Barbara JakubDrilling Method: Air Knife, Hand Auger (HA), and HydropurGeologist: Barbara Jakub / Jeremy QuickSampling Method:Job Number: 38487349.0A022Date(s) Drilled: October 7, 2005 | | | | | | | d Hydropunch (HP) | |
|--|--|---|--|----------------------|-----------|-----------|-------------------|---|
| | BORING IN | FORMA | TION | | | | | |
| Groundwater Depth: Groundwater D | oundwater Not Encountered | Boring | Location: MacA | rthur I | Blvd., | approx. | 100 ft | north of Oakland Ave. |
| Air Knife or Hand Auger | Depth: 12 ft bgs (HA) | Boring | Diameter: 3.25 | " (HA |), 2.75 | 5 " (HP) | | |
| Coordinates: X NA | Y NA | Boring | Type: Explorato | ry HP | Borin | ıg | | |
| Depth (ft bgs) Symbol | Lithologic Descriptio | n | | NSCS | PID (ppm) | Sample ID | Recovery | Comments |
| 0 CON △ GRA GRA GRA | CRETE VELLY SILT: Fill material. Very dark grayi p, 2% clay, 48% silt, 25% sand, 25% angu um plasticity. YEY SILT: Very dark grayish brown (10YR t, 20% clay, 78% silt, 2% fine sand, mediur ft bgs: Color change to dark yellowish bro- ng brown (7.5YR 5/8) and black (10YR 2/1) DY SILT: Yellowish brown (10YR 4/3) with mottling, medium stiff, moist to wet, 3% cla plastic. YEY SILT: Dark grayish brown (2.5Y 4/2) R 4/6) mottling, medium stiff, dry, 15% clay , low to medium plasticity, minor hydrocart 5-8.5 ft bgs: Strong hydrocarbon odor. Y SAND: Brown (10YR 4/3) with grayish to g brown (7.5YR 5/8) mottling, medium der 45% sand, 15% gravel (angular quartz), lo YEY SILT: Brown (10YR 4/3) with grayish to g medium dense, dry, 15% clay, 85% sil- ocarbon odor. | ish brown lar gravel, a 3/2), mee m plasticit own (10Y1) mottling. h grayish y, 82% silt y, 82% silt y, 83% silt y, 93% silt y, 94% silt | (10YR 3/2), low to dium stiff, dry to y. R 4/4) with brown (10YR t, 15% sand, yellowish brown , 2% fine 5Y 5/2) and 5% clay, 25% y, hydrocarbon .5Y 5/2) sticity, | GM ML ML SM | | | | Boring grouted with neat Portland Cement: Top 3-6" finished to grade with cement. |

| UR | | LOG OF BORING | E | Sore | hole l | D:S | B-3 |
|----------------|--------|---|------|-----------|-------------|----------|---------------------------------|
| Depth (ft bgs) | Symbol | Lithologic Description | nscs | PID (ppm) | Sample I.D. | Recovery | Comments |
| - 12 | | SILT: Grayish brown (10YR 5/2) to light yellowish brown (2.5Y 6/3), stiff, dry to moist, 3% clay, 97% silt, minor hydrocarbon odor. HydroPunch driven and exposed from 12 to 14 ft bgs. After 1 hour, no water was available for sampling. HydroPunch driven and exposed from 14 to 16 ft bgs. After 1 hour, no water was available for sampling. HydroPunch driven and exposed from 17 to 19 ft bgs. After 1 hour, no water was available for sampling. | | | | | Bottom of Boring = 19 ft bgs |
| | | | | | | | |

| UCRS 1333 Broadway, Suite 800 Oakland, California 94812 LOG OF BORING PROJECT INFORMATION DRILLING INFORMATION Borland, California 94812 Project IP 41108 Solita Wurker Investigation Drilling Company: Grego Drilling & Techng Site Location: 100 MacAthur Elvel, Oakland, CA Drilling Company: Grego Drilling & Techng Bits Location: 100 MacAthur Elvel, Oakland, CA Drilling Mind Company: Grego Drilling & Techng Bits John MCGain Sampling Mathod: Continuous Manco-Core with acetate sizere. Bits John MCGain Sampling Mathod: Continuous Manco-Core with acetate sizere. Borling Mathod: Drilling & Math M2.5 DP Borling Mathod: Continuous Manco-Core with acetate sizere. Borling II: 20 Abs/s Borling II: 20 Abs/s Borling II: 20 Abs/s Borling II: 20 Abs/s Borling Damster: 2-abn Borling Control Coordinates: X Y Borling Damster: 2-abn Borling II: 20 Abs/s Comments Borling Type: Explanator Borling Damster: 2-abn Borling II: Y SANDY CLAY work Math Baserock (2) Insteams RLL Borling Damster: 2-abn Borling Damster: 2-abn Borling II: Y SANDY CLAY work first Borling, and Saladdy, 10 percellann oddr Borling Damster: 2-abn Borling Damster: 2-abn Borling Baset Abs | | | | <u></u> | | | | | | | | |
|---|-----------------|--|---------------------------------|---|---|---------------------------------------|---------------------------------|---------------|---------|-------------------------|-----------|-----------------|
| Owner Strand California 94612 Benchel D: Sk-4 Total Depth: 32 ft bg Deskel Deskel Project IB #11102 Suit ad Waar Investigation Drilling Company: Greeg Drilling & Testing Project Manager: Lynclic Oxidat Dype of Drilling Rig: Mart M22 DP RG: John McCain Deskel Deskel D: Sk-4 Geologist: Suit Mart Cali Dype of Drilling Rig: Mart M22 DP RG: John McCain Berning In Method: Contract Push Geologist: Suit Mart Cali Deskel Drilling Rig: Mart M22 DP RG: John McCain BORING INFORMATION Groundwater Depth: 20 ft bg Borting Dampier, 2-And Coordinates: X Y Borting Type: Exploratory RG: John McCain Groundwater Object: 5.0 fter bg/filand Auger Borting Type: Exploratory RG: John McCain Groundwater Object: 5.0 fter bg/filand Auger Borting Rig: Mart M22 Stratege Good Bill TY SaMDY CAY vg GRV/EL Fill: Block (1078 27), 756 day, 156 day, | | | | 1222 Broad | hvor Quito Of | 20 | LO | <u>GO</u> | FE | BORIN | <u>IG</u> | |
| Order and the control of the series of the control o | | <u>5</u> - | | Oskland C | iway, Suite of | 10 | Borehole II |): <u>S</u> E | 3-4 | | | |
| PROLECT INFORMATION DRILLING INFORMATION Project: BP #11102 Solit and Water Investigation Drilling Gompany: Gregg Drilling & Texting Bit Location: 100 MacArbur Elw, Oakland, CA Driller. Fase Patilito Project Manager: Lycalle Oxidal Type of Drilling Mitty: Main M.3.5 DP Bit Jocation: Damager: Lycalle Oxidal Damager: Lycalle Oxidal Bit Jocation: Sampling Mathod: Continuous Macro-Core with acture aleve. Jab Number: 34847349.0034 Dete(s) Drilled: 07/1405 Coordinates: X Y Borling Type: Exploratory Coordinates: Coordinates: X Y Borling Type: Exploratory Exploratory Image: State Action of State St | | | | Oakianu, G | | 12 | Total Depti | 1: 32 | ft bj | <u>is</u> | | |
| Project: BP #1102 Sell and Water Investigation Drilling Company: Greeg Drilling & Testing Site Location: 100 MacAnthur Bivd, Oakland, CA. Drilling & Testing & Testing Project: Manager: Lynalls Child Geologite: John McCain Geologite: John McCa | PR | OJE | CT IN | FORMATION | | | DRILI | ING | INFO | DRMAT | ION | |
| Stel Location: 100 Min.Arbur BNd, Okkind, CA Driller: Less Particen Projest Manager: Lynell: Collabi Type of Drilling Rig: Mari M2:5 DP R6: John AC-Lin Define Method: Direr Plad Geologitt: John McCain Sampling Method: Contributus Marce-Core with acetar aleve. Job Number: 39487549.0A034 Dete(s) Drilling Right Coll. Contributus Marce-Core with acetar aleve. Job Number: 39487549.0A034 Dete(s) Drilling Right Coll. Contributus Marce-Core with acetar aleve. Ar Krifs or Hand Auger Depth: 3:0 Beth ges/fand Auger Boring Diameter: 2: Inch Groundwater Depth: 20 B bgs Lithologic Description gg Image: Control of the particle of the state of | Project: BP #1 | 1102 | Soil and | Water Investigatio | n | Driilin | g Company: Gr | egg Di | rilling | & Testin | g | |
| Project Manager: Lynelia Cuitabi RG: John McCain Geologist: Lolan McCain Benjing Method: Continuum Mater-Core with acetars aleve. Job Number: 38487349.0.4034 Detels Diffied: 0714105 Groundwater Depth: 20 ft bgs Groundwater Depth: 20 ft bgs | Site Location: | 100 N | MacArth | ur Blvd, Oakland, C | CA | Driller | : Jesse Pattison | | | | | |
| RG: John McCain Drilling Method: Date: Plab Geologist: John McCain Sampling Method: Continuous Macro-Core with acates slave. Joh Number: 3847249.0A034 Date(s) Drillic: 07/14/05 Borning Leading: Them McCain Borning Leading: States slave. Ar Kinfe or Hand Auger Depth: 30.0 feet bgs/Hand Auger Borning Deather: 2-inch Coordinates: X Y Borning Diameter: 2-inch Commonits Coordinates: X Y Borning Diameter: 2-inch Commonits Geologist: JOI: MCCain States schemeter: 2-inch Coordinates: X Y Borning Diameter: 2-inch States schemeter: 2-inch Coordinates: X Y Borning Diameter: 2-inch States schemeter: 2-inch Commonits Ging Ging Ging Geologist: JOI: MCCain John Diameter: 2-inch States schemeter: 2-inch Coordinates: Interver AC/Baserock: AC cover (S) with baserock (S) banauth Fill Borning Diameter: 2-inch Geologist: JOI: MCCain Gin Bit Diameter: 2-inch Gin Bit Diameter: 2-inch Geologist: JOI: MCCain Gin Bit Diameter: 2-inch States schemeter: 2-inch Geologist: JOI: MCCain Gin Bit Diameter: 2-inch Gin Bit Diameter: 2-inch Juit Diameter: Actin Bit Diameter | Project Manag | er: ľ | ynelle C | nishi | | Туре с | of Drilling Rig: N | Aarl M | 2.5 D | P | ****** | |
| Geologiet: John McCain Sampling Method: Continuous Macro-Care with acetate slave. Jdb Number: 348(7349.0034 Data(s) Drilled '07/14/05 BORING INFORMATION Groundwater Depth: 20 ft bgs Boring Incodent: Scance of site Ar Knife or Hand Auger Depth: 5.0 fteet bgs/Hand Aager Boring Type: Explorator: Scance of site Coordinates: X Y Boring Type: Explorator: Scance of site Good Affect of Hand Auger Depth: 5.0 fteet bgs/Hand Aager Boring Type: Explorator: Y Good Affect of Hand Auger Depth: 5.0 fteet bgs/Hand Aager Boring Type: Explorator: Y Good Affect of Hand Auger Depth: 5.0 fteet bgs/Hand Aager Boring Type: Explorator: Y Good Affect of Hand Auger Depth: 5.0 fteet bgs/Hand Aager Boring Type: Explorator: Y Good Affect of Hand Auger Depth: Soft of Hand Auger Dep | RG: John McCai | in | | • | | Drillin | g Method: Direc | t Push | | | | |
| Job Number: 33487349.04034 Date(s) Drilled: 07/14/05 BORING INFORMATION Groundwater Depth: 20 ft bgs Boring Location: SE comer of site Arr Knife or Hand Auger Depth: 5.0 feet bgs/Hand Auger Bering Location: SE comer of site Coordinates: X Y Boring Type: Exploratory Image: Second Colspan="2">Goordinates: X Y Boring Type: Exploratory Goordinates: X Y Boring Location: Site comer of site Comments Site Second Comer of Site Sec | Geologist: John | n Mc | Cain | | | Samp | ing Method: Co | ntinuo | us M | acto-Core | e with a | icetate sleeve. |
| BORING INFORMATION BORING INFORMATION Boring Leadion: Sizement of size Air Knife or Hand Auger Dapth: 5.0 feet bgs/Hand Aager Boring Leadion: Sizement of size Coordinates: X Y Boring Leadion: Sizement of size Coordinates: X Y Boring Leadion: Sizement of size Image: Size of the Size of Size | Job Number: 3 | 38487 | 7349.0A(|)34 | | Date(s |) Drilled: 07/14/ | 05 | | | | |
| Borning Location: Site comer of site Air Knife or Hand Auger Depth: 5.0 feet bgs/Hand Auger Boring Liceation: Site comer of site Coordinates: X Y Boring Type: Exploratory Image: Site of the state of site Site of the state of site Site of the state of site Image: Site of the state of the state of site Site of the state of the state of site Image: Site of the state of | | | | | BORING IN | FORMA | TION | | | | | |
| Arr Knite or Hand Auger Boring Diameter: 2-Inch Coordinates: X Y Boring Type: Exploratory Image: State of the state of | Groundwater | Depti | h: 20 ft | bgs | | Boring | Location: SE co | mer o | fsite | | | |
| Coordinates: X Y Boring Type: Exploredory Image: Second Sec | Air Knife or Ha | and A | Auger D | epth: 5.0 feet bgs/ | Hand Auger | Boring | Dlameter: 2-inc | h | | | - | |
| Bit Comments Comments 0 AC/Gasence: AC cover (*) with basence (3*) baneath FILL 1 SILTY SANDY CLAY w/ GRAVEL: FILL, Mark (10YR 2/1), 76% day, for invasion of the transfer of the transfer of the covers sense, more inputs of the covers input sense input | Coordinates: | X | | Y | | Boring | Type: Explorate | хy | | | | |
| En Luccourse Description State B State Control of R 0 AC/Basenock: AC cover (6") with basenodk (3") baneath frames and block frequencies of anneath frames and block frequencies of anneath poperoisum odor FILL Sultry SANDY CLAY: or GRAVEL: FILL, black (10YR 2/1), 75%, dep, frames and block frequencies of anneath poperoisum odor FILL gravels and block frequencies of anneath problem odor FILL gravels anneath gravels gravels anneath gravels FILL gravels gravels anneath gravels FILL gravels gravels gravels gravels gravels FILL gravels gravels gravels gravels gravels FILL gravels gravels gravels gravels FILL gravels gravels gravels gravels gravels FILL gravels gravels gravels gravels gravels FILL gravels g | (ft bgs) | 8 | | 1 14 | hologic Decedet- | | | S | (mqq | le ID | very | Commonte |
| 0 AC/Baserock: AC cover (6°) with baserock (3°) beneath FILL 2 SILTY SANDY CLAY w(GRAVEL: FILL, black (10YR 27), 7% day, 19% d | Depth | eym Sym | | | | 11 | | ns: |) (IId | Samp | Rec | Continents |
| 2 SiLTY SANDY CLAY w GRAVEL: FILL, black (10YR 2/1), 75% clay, 15% send, 7% clay, 15% send, 5% clay, 15% send, 10% send, | E O | | AC/Bas | erock; AC cover (6* | ") with baserock (3") | beneath | | FILL | | | | |
| 2 15% send, 5% git, 5% git, 2% gravel, fine to coarse sends, race angular gravels and brick tegramets to 2* diameter, soft, molst, molst, molst, molst, gits staticity, no petroleum odor grave and brick tegramets to 2* diameter, soft, molst, molst, molst, molst, fine sends, fine se | | | SILTY | SANDY CLAY W/ GI | RAVEL: FILL, black (| 10YR 2/1 |), 75% clay, | | | Borahola | | |
| 10 Bit TY SANDY CLAY: dark brown (10YR 3/3), 75% day, 15% send, 5% CL 0 gades with Portland carnet 6 Image: Sint Sint Sint Sint Bit Distances and Sint Sint Sint Bit Distances and Sint Sint Sint Bit Sint Bit Distances and Sint Sint Bit Distances and Sint Bit Sint Bit Distances and Sint Bit Bit Distances and Sint Bit Distances and Sint Bit Bit Bit Distances and Sint Bit Bit Bit Bit Bit Bit Bit Bit Bit Bi | -2 | | 15% sa gravel a | and, 5% slit, 5% grav and brick fregments | vel, fine to coarse sai to 2* diameter, soft, | nds, trace moist, me | angular d. plasticity, | | | grouted to | | |
| 4 SILTY SANDY CLAY: dark brown (10YR 3/3), 75% day, 15% send, 5% CL 0 Stad 6 Stad Stad CL 0 Stad 6 G - same as above, no gravel, no petroleum odor 0 Stad Stad 7 Stad Stad Stad Stad Stad 6 G - same as above, no gravel, no petroleum odor 0 Stad Stad Stad 10 G - same as above, no gravel, fine to correspond to 0.25°, molet, med. stad 0 Stad Stad Stad 10 G 10° - Sity Sandy Clay continues, reddsh-brown, (SYR 5/4), 70%, clay, 10% stit, 10% sand, 5% gravel, fine to correspond sends, trade sub-rounded gravels to 0.25°, molet, med. stift, no petroleum odor 0.2 Stad Stad 11 G 15° - Sity Sandy Clay continues, dark reddish gray (2.5YR 4/2), S0% clay, 10% stit, 10% sand, no gravels, fine sends, molet, med. 0.2 Stad Stad 116 StLTY SAND: brown (7.5YR 5/3), 90% samd, 10% stit, fine sends, bose, wet, no petroleum odor 0.4 Stad St | | | ño petr | oleum odor | | • | | | | grade with | | |
| 4 SILTY SANDY CLAY: dark brown (10YR 33), 75% sign, 15% send, 5% clay, 15% send, 100, 5% CL 0 SB-4, (6.5.5) 6 @ 6' - same as above, no gravel, no petroleum odor 0 SB-4, (6.5.5) SB-4, (6.5.5) 10 @ 10' - Silty Sandy Clay continues, raddish-brown, (5YR 5/4), 70% clay, 10% sint, 10% sint, 10% sint, 5% gravel, fine to coarse sands, trade sub-vounded gravels to 0.25°, molat, med. stift, no petroleum odor 0.2 SB-4, (6.5.7) 112 0 15 - Silty Sandy Clay continues, dark reddish gray (2.5YR 4/2), 80% day, 10% sint, 10% sint, med. stift, no petroleum odor 0.2 SB-4, (14.5.16) 114 @ 15 - Silty Sandy Clay continues, dark reddish gray (2.5YR 4/2), 80% day, 10% sint, no petroleum odor 0.2 SB-4, (14.5.16) 116 stift, no petroleum odor 1.9 SB-4, (14.5.16) 1.9 118 SILTY SANDY CLAY: brown (7.5YR 5/3), 90% send, 10% silt, fine sends, hold, med. stift, molat to wel, no petroleum odor, med. stift, fine sends, fine s | E, | | | | | | | | | Portland | | |
| 6 @ 6' - same as above, no gravel, no petroleum odor 02 SB-4 (0.5-0) 10 @ 10' - Silty Sandy Clay continues, reddish-brown, (5YR 5/4), 70% clay, 10% silt, 10% sand, 5% gravel, fine to coarse sands, trace sub-rounded gravels to 0.25°, molet, med. stiff, no petroleum odor 02 SB-4 (0.5-10) 11 @ 15' - Silty Sandy Clay continues, dark reddish gray (2.5YR 4/2), S0% clay, 10% silt, 10% sand, no gravels, fine sands, molst, med. 0.2 SB-4 (14.5-17) 16 stiff, no petroleum odor 1.0 SB-4 (14.5-17) 0.2 18 g. 15' - Silty Sandy Clay continues, dark reddish gray (2.5YR 4/2), S0% clay, 10% silt, 10% sand, no gravels, fine sands, molst, med. 1.0 SB-4 (14.5-17) 18 9.5 SB-4 (16.5-20) SB-4 (16.5-20) SB-4 (16.5-20) 20 SILTY SANDY CLAY; brown (7.5YR 5/3), 70% clay, 15% silt, 15% sand, fine sands, med, stiff, moist to wel, no petroleum odor 9.5 SB-4 (20.20.5) SB-4 (20.20.5) 22 SILTY SANDY CLAY; brown (7.5YR 5/3), 70% clay, 15% silt, 15% spasid(t) CL plasticity Borehole ID : SB-4 | -4 | | SILTY silt, 5% diamete | SANDY CLAY: dark gravel, fine to coan ar, soft, molst, med. | brown (10YR 3/3), 7 se sands, trace sub- plasticity, no petrole | '5% clay, ' angular gri um odor | 15% sand, 5% avel to 0.5" | CL | o | SB-4 | | |
| 8 0.2 SB-4 (0.5-10) 10 (0.10 ⁺ - Silly Sandy Clay continues, reddish-brown, (5YR 5/4), 70% Clay, 10% silt, 10% sand, 5% gravel, fine to coarse sends, frace sub-rounded gravels to 0.25 ⁺ , molst, med. stiff, no petroleum odor 0.2 SB-4 (0.5-10) 12 (0.10 ⁺ - Silly Sandy Clay continues, dark reddish gray (2.5YR 4/2), sub-rounded gravels, bit 0.25 ⁺ , molst, med. stiff, no petroleum odor 0.2 SB-4 (14.5-15 ⁺) 16 9.0 ⁺ O - Silly Sandy Clay continues, dark reddish gray (2.5YR 4/2), stiff, no petroleum odor 0.2 SB-4 (14.5-15 ⁺) 18 9.0 ⁺ Sill, TY SAND: brown (7.5YR 5/3), 90% sand, 10% silt, fine sands, loose, wet, no petroleum odor 0.8 SB-4 (20.20.5 ⁺) 1.9 20 Sill, TY SAND: brown (7.5YR 5/3), 90% sand, 10% silt, fine sands, loose, wet, no petroleum odor SM 0.8 SB-4 (20.20.5 ⁺) SB-4 (20.20.5 ⁺) 21 Sill, TY SANDY CLAY: brown (7.5YR 5/3), 70% clay, 15% silt, 15% sand, fine sands, med. stiff, moist to wet, no petroleum odor, med. plasticity SM Cl. SB-4 22 Sill, TY SANDY CLAY: brown (7.5YR 5/3), 70% clay, 15% silt, 15% sand, fine sands, med. stiff, moist to wet, no petroleum odor, med. plasticity Cl. Borehole ID : SB-4 | 6 | | @ 6' - 5 | ame as above, no (| gravel, no petroleum | odor | | | | (0-0.07) | | |
| 10 @ 10' - Sitty Sandy Clay continues, reddish-brown, (5/R 5/4), 70% [6.5-10'] 12 | 8 | a a sua da faranza angle Aliang a sua angle Aliang Sua anglesian | | | | | | | 0.2 | SB-4 | | |
| 14 @ 15 - Silty Sandy Clay continues, dark reddish gray (2.5YR 4/2), 80% clay, 10% silt, 10% sand, no gravels, fine sands, moist, med. 0.2 SB-4 (14.5-15) 16 stiff, no petroleum cdor 1.9 SB-4 (14.5-15) 18 SiltTY SAND: brown (7.5YR 5/3), 90% sand, 10% silt, fine sands, cose, wet, no petroleum cdor 9.6 SB-4 (20-20.5) 20 SiltTY SAND: brown (7.5YR 5/3), 90% sand, 10% silt, fine sands, cose, wet, no petroleum cdor SM 9.6 SB-4 (20-20.5) 21 SiltTY SANDY CLAY: brown (7.5YR 5/3), 70% clay, 15% silt, 15% sand, fine sands, mod. stiff, moist to wet, no petroleum cdor, med. plasticity CL Borehole ID : SB-4 | - 10 | | @ 10' - clay, 10 sub-rou | Silty Sandy Clay co 1% silt, 10% sand, 5 Inded gravels to 0.2 | ntinues, reddish-bro % gravel, fine to coa 5°, moist, med. stiff, ; | wn, (5YR : ree sands no petrole | 5/4), 70% , trace um odor | | | (9.6-10') | | |
| 18 1.9 SB-4 (19.5-20) 20 SILTY SAND: brown (7.5YR 5/3), 90% sand, 10% slit, fine sands, loose, wet, no petroleum odor 9.6 22 SILTY SANDY CLAY: brown (7.5YR 5/3), 70% clay, 15% slit, 15% sand, fine sands, med. stiff, moist to wet, no petroleum odor, med. 0.6 24 SILTY SANDY CLAY: brown (7.5YR 5/3), 70% clay, 15% slit, 15% sand, fine sands, med. stiff, moist to wet, no petroleum odor, med. CL 24 BP/Atlantic Richfield Company Page 1 of 2 | 14 | | @ 15' - 80% cla stiff, no | Silty Sandy Clay co ay, 10% silt, 10% sa petroleum cdor | ontinues, dark reddiel Ind, no gravels, fine s | h gray (2.5 sands, mo | YR 4/2), ist, med. | | 0.2 | SB-4 (14.5-15) | | |
| 20 SILTY SAND: brown (7.5YR 5/3), 90% sand, 10% slit, fine sands, loose, wet, no petroleum odor SM 9.6 SB-4 (20-20.5) 22 SILTY SANDY CLAY: brown (7.5YR 5/3), 70% clay, 15% slit, 15% sand, fine sands, med. stiff, moist to wet, no petroleum odor, med. plasticity CL CL BP/Atlantic Richfield Company Page 1 of 2 Borehole ID : SB-4 | 118 | | | | | | | | 1.9 | SB-4 (19.5-20') | | ~ |
| E. SILTY SANDY CLAY: brown (7.5YR 5/3), 70% clay, 15% silt, 15% sand, fine sands, med. stiff, moist to wet, no petroleum odor, med. CL Page 1 of 2 Borehole ID: SB-4 | 20 | | SILTY loose, v | SAND: brown (7.6Y) wet, no petroleum o | R 5/3), 90% sand, 10 dor |)% siit, fina | e sends, | SM | 9,6 | (20-20.5 [°]) | | |
| BP/Atlantic Richfield Company Page 1 of 2 Borehole ID : SB-4 | - 24 | | SILTY sand, fi plastici | SANDY CLAY: brow ine sands, med. stiff ty | m (7.5YR 5/3), 70% (, moist to wet, no pe | clay, 15% troleum oc | silt, 15% ior, med. | CL | | | | |
| | BP/Atla | anti | c Rici | nfield Compa | any Page 1 | of 2 | | | E | Iorehol | e ID : | SB-4 |

| UNG | LOG OF BORING | OF BORING Borehole ID | | | |): SB-4 | | |
|----------------------------------|---|-----------------------|-----------|--|----------|----------|--|--|
| Depth (ft bgs) Symbol | Lithologic Description | uscs | (mqq) Olq | Sample I.D. | Recovery | Comments | | |
| 26 28 30 32 34 34 | @ 25' - Siliy Sandy Clay (CL) continues, light brown (7.5YR 8/3), 70% clay, 15% silt, 15% sand, med. stiff, molst to wet, no petroleum odor @ 26' - Silty Sandy Clay continues, color change to gray (Giey 1 5/10Y) at 28', no petroleum odor SILTY SAND: brown (7.5YR 5/3), 90%sand, 10%silt, fine sands, loose, wet, no petroleum odor SILTY SANDY CLAY: gray (Gley 1 5/10Y), 70% clay, 15% silt, 15% sand, fine sands, stiff, molst, no petroleum odor, med. plasticity SILTY SANDY CLAY: gray (Gley 1 5/10Y), 70% clay, 15% silt, 15% sand, fine sands, stiff, molst, no petroleum odor, med. plasticity SILTY SANDY CLAY: gray (Gley 1 5/10Y), 70% clay, 15% silt, 15% sand, fine sands, stiff, molst, no petroleum odor, med. plasticity SILTY SANDY CLAY: gray (Gley 1 5/10Y), 70% clay, 15% silt, 15% sand, fine sands, stiff, molst, no petroleum odor, med. plasticity SILTY SANDY CLAY: gray (Gley 1 5/10Y), 70% clay, 15% silt, 15% sand, fine sands, stiff, molst, no petroleum odor, med. plasticity INTY SANDY CLAY: gray (Gley 1 5/10Y), 70% clay, 15% silt, 15% sand, fine sands, stiff, molst, no petroleum odor, med. plasticity INTY SANDY CLAY: gray (Gley 1 5/10Y), 70% clay, 15% silt, 15% sand, fine sands, stiff, molst, no petroleum odor, med. plasticity | SM CL SM CL | 0.5 | SB-4 (25-25.57) SB-4 (29-29.57) | | | | |

| PRO Project: Former Site Location: 10 Project Manage PG: Barbara Jakub Geologist: Jeren Job Number: 38 | DJECT INFO BP #11102 00 MacArthur r: Lynelle Or my Quick 4487349.0A02 | 1333 Broadw Oakland, Calif ORMATION r Boulevard, Oakland nishi | ay, Suite 80 fornia 9461 , CA BORING INF | LOG OF BORING Jite 800 Borehole ID: SB-4A 94612 Total Depth: 36 ft bgs DRILLING INFORMATION DRILLING INFORMATION Drilling Company: Gregg Drilling & Testing Driller: Paul Rogers Type of Drilling Rig: MARL M10T Drilling Method: Airknife and Direct Push Technology (DP) Sampling Method: Soil: Acetate by MacroCore; Groundwater: 3 V Date(s) Drilled: October 7, 2005 ING INFORMATION Date of Drilled: October 7, 2005 | | | | | | |
|--|---|---|--|--|--|----------------|-----------|---|----------|---|
| Groundwater De Air Knife or Han | epth: 24.5 ft d Auger Dep | bgs oth: 5.2 ft bgs (HA) | | Boring Boring | Location: South Diameter: 2.7: | nwest o | orner | , on-site n | ear foi | mer boring SB-4. |
| Coordinates: | X NA | Y NA | | Boring | Type: Explorate | огу | | | | |
| Depth (ft bgs) | loamye | Litho | ogic Descriptio | n | | nscs | PID (ppm) | Sample ID | Recovery | Comments |
| | ASPHAL CLAYEY dense, d Notable CLAYEY damp, 19 gravel, m @ 4-5.2 @ 5.2-6 medium @ 6-9 ft plasticity staining. SAND: G 4/2) motil course sa depth. | T SILTY SAND: Yellow amp, 10% clay, 30% s oxidation staining. SANDY SILT: Dark t % clay, 70% silt, 25% CLAYEY SILT: Gravis % clay, 80% silt, 5% edium to high plasticit ft bgs: Notable oxidat ft bgs: Color change plasticity. Notable oxid bgs: Color change to . Weathered granite of ireenish gray (GLEY ling, loose to medium and, non-plastic, hydr | vish brown (10YR silt, 60% fine sand prown (10YR 3/3) sand, medium pla sh brown (10YR 3/3) sand, medium pla sh brown (10YR 5/ dation staining. to gray (10YR 5/ dation staining. brown (10YR 5/ clasts throughout. 1 5/5GY) with dat dense, moist, 2% ocarbon odor. In | 5/4), med d, medium , soft to m asticity. 5/2), very : im angular 1), mediur 3), low to t Notable of Notable of rk grayish day, 8% s creasing f | lium dense to plasticity. edium stiff, soft to soft, r white chert n stiff, medium prown (10YR silt, 90% fines with | SM ML SM | 3.1 | SB-4A -6' 09:55 SB-4A -10' 10:13 | | Boring grouted with neat Portland Cement. Top 3-6" finished to grade with cement. Airknife could not penetrate, hand auger used instead. Top 5.2 feet of boring logged from hand auger cuttings. |
| | 20% grav | 5iLT: Brown (10YR 5/ /el, low plasticity. | 3), ɗamp, 3% clay | /, 47% silt | , 30% sand, | ML | | | | |

Borehole ID: SB-4A

| UI | R | LOG OF BORING | I | Bore | ehole | ID : S | SB-4A |
|---|------------|--|----------|-----------|--|----------|--|
| Depth (ft bgs) | Symbol | Lithologic Description | USCS | PID (ppm) | Sample (.D. | Recovery | Comments |
| - 12 | <u> </u> : | | <u> </u> | | | | |
| - - - - - - - - - - - - - - - - - - - | | No recovery from 12 to ~15.8 ft bgs, Could not remove acetate liner from Macrocore sampler. | | | | | |
| - 1 6 | | CLAYEY SILT: Pale brown (10YR 6/3), medium stiff to stiff, damp, 5% clay, 95% silt, low plasticity. Logged from sampler shoe. SAND: Dark yellowish brown (10YR 4/4) with dark grayish brown (10YR 4/2) mottling and very dark gray (10YR 3/1) patches, damp, loose to medium dense. 2% clay. 8% silt. 80% sand 10% ~2-3 mm grayel | ML SP | 9.1 | | | |
| - 18 | | slight hydrocarbon odor. Notable oxidation staining. @ 16.2-17 ft bgs: Grayish brown (10YR 5/2) with gray (10YR 5/1) patches, CLAYEY SILT: Gray (10YR 5/1) to dark grayish brown (10YR 4/2), medium stiff, damp, 15% clay, 80% silt, 5% sand, trace gravel (likely scrape material), low to medium plasticity, slight hydrocarbon odor. | ML. | 7.6 | | | |
| - 20 | | @ 20-22 ft bgs: Color change to yellowish brown (10YR 5/4). Soft to medium stilf. Hydrocarbon odor. | | 21.5 | SB-4A -20' 10:23 | | |
| - 22 | | @ 22-23 ft bgs: Increasing fines, no sand. Hydrocarbon odor. | | 28.7 | | | |
| | | SILTY GRAVELLY SAND: Light olive brown (2.5Y 5/4), loose to medium dense, damp, 1% (minor) day, 19% slit, 60% sand, 20% gravel, non-plastic. CLAYEY SILT: Light olive brown (2.5Y 5/3), soft to medium stiff, damp. | SP | | | | |
| - 26 | | 10% clay, 90% silt, low to medium plasticity, slight hydrocarbon odor. | | 3.2 | SB-4A 12:05 SB-4A -25' 10:44 | | Boring was initially dry. Groundwater elevation measured after water was allowed to accumulate in the open boring for more thorage |
| | | @ 27.5-30 ft bgs: Color change to grayish brown (2.5Y 5/2) with dark gray (2.5Y 4/1) mottling to 28 ft bgs. Stiff to very stiff, low plasticity. | | 0.9 | | | hour. |
| P/Atlan | tic Ri | chfield Company Page 2 of 3 | | Bo | rehole | ייתו | SR-44 |

.

| UR | | LOG OF BORING | E | lore | hole l | D:SI | B-4A |
|----------------|--------|--|------|-----------|------------------------|----------|---------------------------------|
| Depth (ft bgs) | Symbol | Lithologic Description | nscs | PID (ppm) | Sample I.D. | Recovery | Comments |
| - 30 | | @ 30-36 ft bgs: Color change to dark grayish brown (10YR 4/2). Stiff to very stiff. | | 5.2 | SB-4A -30' 10:49 | | |
| - 34 | | @ 35-36 ft bgs: Light gray (10YR 7/1) mottling. Very stiff. | | 0.8 | SB-4A -35' 10:58 | | Bottom of Boring = 36 ft bgs |
| | | • | | | | | |
| | | | | | | | |
| | · | | | | | | |

......

Borehole ID: SB-4A

| TIRS | 1333 Broadway, Suite 8 | LOC Borehoie ID | GO : SB | F E | BORIN | IG | 1999 a Angelon and an | |
|--|---|---|--|--------|-------------------|--|---|-----------------|
| VIW | Oakland, California 946 | 12 | Total Depth | : 32 | ft bg | js | | |
| PROJECT I | NFORMATION | | DRILL | ING I | NFC | ORMAT | ION | |
| Project: BP #11102 Soil a | nd Water Investigation | Drillin | g Company: Gre | gg Dr | illing | & Testin | g | ···· |
| Site Location: 100 MacAr | thur Blvd, Oakland, CA | Driller: Jesse Pattison | | | | | | |
| Project Manager: Lynelle | Onishi | Type of Drilling Rig: Marl M2.5 DP | | | | | | |
| RG: John McCain | | Drillin | g Method: Direct | Push | | | | |
| Geologist: John McCain | | Sampl | ing Method: Cor | tinuo | us M | acro-Core | with a | icetate sleeve. |
| Job Number: 38487349.0. | A034 | Date(s |) Drilled: 07/14/0 |)5 | | | | |
| | BORING IN | FORMA | TION | | | | | · · · |
| Groundwater Depth: 29: | ft bgs | Boring | Location: Southy | vest o | f disp | enser isla | mds | |
| Air Knite or Hand Auger | Depth: 5.0 feet bgs/Hand Auger | Boring | Diameter: 2-incl | 1 | | | | |
| Coordinates: X | Υ | Boring | Type: Explorator | У | | | | |
| Depth (ft bgs) Symbol | Lithologic Description | n | | nscs | PID (ppm) | Sample ID | Recovery | Comments |
| О АС/В | aserock: AC cover (6") with baserock (2") | beneath | | FILL | | | | ····· |
| SANI grave soft, 1 grave soft, 1 grave stift, 1 grave stight | UY GLAY: FILL, black (Giey 1 2.5/N), 80% al, fine to coarse sands, trace angular gravmoist, low plasticity, no petroleum odor - Sandy Clay FILL continues, greenish greday, 15% sand, soft, molst, low plasticity, - same as above, gray (Gley 2 4/5BG), angrete fragments to 3° diameter, soft, molst, soft, molst, soft, molst, low plasticity, rose automatication of the same as above, gray (Gley 2 4/5BG), angrete fragments to 3° diameter, soft, molst, soft, molst, soft, molst, invest, soft, molst, soft, molst, low plasticity, no state fragments to 3° diameter, soft, molst, soft, molst, soft, molst, invest, soft, molst, soft, gravel, fine sands, trace sub-rounded grades, med. plasticity, no petroleum odor Y - Sility Sandy Clay continues, brown (7.5Y slit, 10% sand, 5% gravel, trace angular gradeter, med. stiff, molst, med. plasticity, no petroleum odor Y - Sility Sandy Clay continues, brown (7.5Y slit, 10% sand, 5% gravel, trace angular gradeter, med. stiff, molst, med. plasticity, no petroleum odor Y - Sility Sandy Clay continues, brown (7.5Y slit, 10% sand, 5% gravel, trace angular gradeter, med. stiff, molst, med. plasticity, no petroleum odor Y - Sility Sandy Clay continues, brown (7.5Y slit, 10% sand, 5% gravel, trace angular gradeter, med. stiff, molst, med. plasticity, no petroleum odor | clay, 15% els to 0.25 y (Gley 1 f no petrolei gular grave slight petrol clay, 10% ravels to 0.25 avel to 0.25 etroleum c ravels to 0.25 etroleum c | sand, 5% ' diameter, W5GY), Im odor Ms and teum sand, 10% 25", med. % clay, 5" dor % clay, 5" dor R 5/4), ticity, | CL | 0.0 0.1 0.4 | Borehole grouted to grade with Portland cement SB-5 (5-5.5°) SB-5 (9.5-10°) SB-5 (9.5-10°) SB-5 (14.5-15°) SB-5 (19.5-20°) | | |
| 1 24 1 1 1 1 1 1 1 1 1 1 | · · · · · · · · · · · · · · · · · · · | | | | | | | |
| BP/Atlantic Rid | chfield Company Page 1 | of 2 | | | B | lorehold | e ID : | SB-5 |
| UR | | LOG OF BORING | E | Bore | hole I | D:S | B-5 |
|----------------|--|---|-----------|-----------|--------------------|----------|----------|
| Depth (ft bgs) | Symbol | Lithologic Description | USCS | PIO (ppm) | Sample I.D. | Recovery | Comments |
| 26 | osta algund "and" 1944 - Alexandro Alexandro Alexandro 1944 - Alexandro Alexandro Alexandro Alexandro Alexandro Alexandro Alexandro Alexandro Alexandro Alexandro Alexandro | @ 25' - Silty Sandy Clay (CL) continues, brown (7.5YR 5/3), 78% clay, 10%silt, 10% sand, 2% gravel, trace gravel to 0.25" diameter, med. stiff, moist, med. plasticity, no petroleum odor | | | | | |
| 30 | | SILTY SAND: brown (7.5YR 5/3), 85% sand, 10% slit, 5% clay, fine to coarse sands, loose, wet, no petroleum odor SILTY SANDY CLAY: brown (7.5YR 5/3), 85% clay, 10% slit, 5% sand, fine to coarse sands, stiff, moist, med. plasticity, no petroleum odor | SM/ CL | 0.4 | SB-5 (29-29.5') | | × |
| 34 | | Depth discrete groundwater samples were attempted within a boring 1 foot laterally from this location and were not successful. | | | | | |
| | | | | | | | |
| | | | | | | | |

| | 3 | C. | 4000 D | | | LC | GC |)F I | BORI | NG | | . <u> </u> |
|---|--|----------------------|--|--|------------------------------|-----------------|-----------|---------------|---|-------------|-----------------|------------|
| | N. | | 1333 Broad | Nay, Suite 8 | 00 | Borehole | ID: S | B-6 | | | | |
| | | | Carianu, Ca | inomia 940 | 12 | Total Dept | th: 2 | 8 ft b | gs | | | |
| P | PROJECT INFORMATION DR | | | DRIL | LLING INFORMATION | | | | | | | |
| Project: BP #11102 Soil and Water Investigation | | | Drillin | g Company: G | regg I | Drillin | g & Testi | ıg | | | | |
| Project Mana | : 100 | MacAnth | ur Blvd, Oakland, CA | \ | Driller | Jesse Pattison | | | | | | |
| RG: John McC | gen.) ain | Lynene U | <u>nisni</u> | | Туре с | f Drilling Rig: | Marl N | <u>12.5 I</u> |)P | | | |
| Geologist: Jol | hn Mr | cCein | | | Drillin | g Method: Dire | ct Pus | h | | | | |
| Job Number: | 3848 | 7349.0A0 |)34 | | Data/a | | ontinu | ous M | lacto-Con | e with | acetate sleeve. | |
| | | | ···· | BORING IN | FORMA | | 5/05 | | | | | |
| Groundwater | Dept | th: 8.5 ft | bgs | Derand In | Boring | Location: Bety | veen st | ation l | | nd dia | | |
| Alr Knife or H | land / | Auger D | epth: 5.0 feet bgs/H | and Auger | Boring | Diameter: 2-in | ch | | Junumy a | no uis | pensers | |
| Coordinates: | X | | Y | 0 | Boring | Type: Explorat | OTY | | | •• | | |
| <u>.</u> | | | | | 1 | 2F-13- | <u> </u> | 1 | 1 | | 1 | |
| bgs | | | | | | | | F | <u> </u> | 2 | | |
| ц Ц | đ | | Lithe | ologic Description | n | | 1 20 | l ā | ble | e Ag | Comments | |
| Jept | Ś | | | | | | ្រី | l d | am | Rec | | |
| | | | | | | | | | 0) | | | |
| Ē | | AC/Bas | erock: AC cover (3") | with baserock (4*) | beneath | | FILL |] | Ţ | | | |
| Ē | | SILTY S | SAND: FILL, red (2.5) | R 5/6), 85% sand, | 10% slit, 5 | 5% gravel, | 1 | - | Borehoje | | | |
| -2 | 主臣 | petroleu | m odor | reis in U.Z. Gianne | 10088, | | | | grouted to | | | |
| | | | | | | | | | with | | | |
| E-4 | | | | | | | | | Portland | | | |
| | | | | | | | | | | | | |
| | | | | | | | | 0 | SB-6 (5-5 5) | 190.Z | | |
| – 6 | | | Ma. A. F. H. | | | | | | (00.07 | | | |
| | 記述 | to 0.5" d | lity Sand continues, s liameter, loose, moist | ame color as abov , no petroleum odo | re, angular Ir | gravela | | | | | | |
| -8 | | @ 8.5' - petroleu | Slity Sand continues, modor | red (2.5YR 5/6), id | bose, wat a | it 8.5', | | | | 1000 | | |
| | 繁璧 | 011 774 0 | | | | | | 289 | SB-6 (8.5.01) | | | Z |
| E-10 | | sand, 15 | 5% silt, 5%, fine sand: | eyish brown (10YR s, med. stiff, molst, | k 4/2), 75% , no petroie | um odor | CL | 0.9 | SB-6 | | | |
| Ē | | | | | | | | | (9.5-10') | | | |
| ŧ | | | | | | | | | | | | |
| F 12 | | | | | | | | | | | | |
| | | | | | | | | | | | | |
| - 14 | | | | | | | | | | | | |
| E I | -22- | SILTY C | LAYEY SAND: brown | (10YR 5/3), 75% | sand, 10% | silit, 10% | SM | 0.4 | SB-6 (14.5-15 ⁵) | 金橋 | | |
| 16 | <u>>></u> | wet, no p | petroleum odor | ~ ୦୦.୮୦୬, ଖାସ୍ତ୍ରପାଥୀ ପ୍ର | paveis to U | .27", 19098; | | | , | | | |
| E | <u>~</u> | | | | | | | 0.6 | SB-6 | | | |
| Ē | | SILTY S | ANDY CLAY: dark ye | llowish brown (10Y | (R 4/4), 80 | % clay, 10% | CL | | (18.5-17) | | | |
| 18 | | odor, me | id. plasticity | | noren no bi | | | | | | | |
| | | | | | | | | | 2008 - 1941 - 1942 - 1942 - 1942 - 1942 - 1942 - 1942 - 1942 - 1942 - 1942 - 1942 - 1942 - 1942 - 1942 - 1942 - | | | |
| <u>–</u> 20 | | @ 20' - 5 | Silty Sandy Clay conti | rues, vellowish bro | Wm (10YR | 5/4) | | 0.0 | SB-6 (19.6-20) | | | |
| | - trace angular gravel to 0.25" diameter, med. stiff, molet, no odor | | | | | | | | | | | |
| 22 | | | | | | | | | | | | |
| | | Clay con | ery πom 20 - 24° push tinues, yellowish brow | n, soil in shoe @ 24 m (10YR 5/4), fine | 1'- Silty Sar to coarse a | ndy sands, | | | | | | |
| Ē | med. stiff, med. plasticity, slight petroleum odor | | | | | | | | | | | |
| <u>1-24</u> | | | | | | | í [| ł | <u> </u> | | | |
| BP/Atia | Intic | c Kich | neld Company | y Page 1 | of 2 | | | B | orehole | ID : | SB-6 | |

| UR: | LOG OF BORING | E | Bore | hole | D:S | B-6 |
|------------------------------|---|------|-----------|--------------------------------|----------|-------------|
| Depth (ft bgs) Symbol | Lithologic Description | uscs | PID (ppm) | Sample I.D. | Recovery | Comments |
| - 26 - 28 - 30 - 32 | Silty Sandy Clay (CL) continues, grayish brown (10YR 5/2), fine sands, stiff, med. plasticity, no petroleum odor Bottom of Boring= 28' bgs Depth discrete groundwater samplas were attempted within a boring 1 foot laterally from this location and were not successful. | | 0,0 | SB- 0 (27.5-28") | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| RP/Atlantic | Richfield Company Bara 2 of 2 | | | rahala | ID • • | TP 6 |

| TTDC | 1333 Broadway Suite P | nn | LO | GO | FE | BORIN | NG | |
|----------------------------------|--|--|--------------------------------|---------|-----------|---------------------------|----------|---------------------------------------|
| UND | Oakland, California 94612 | | Borehole II | D: SE | 3-7 | <u> </u> | | · · · · · · · · · · · · · · · · · · · |
| | | , | I otal Liept | n: 32 | ft bį | <u>8</u> 8 | | |
| PROJECT IN | IFORMATION | ļ | DRILI | ling | INFO | DRMAT | ION | |
| Project: BP #11102 Soil er | ad Water Investigation | Drilling | Company: Gr | egg Di | illing | & Testin | ıg | |
| Site Location: 100 MacArt | nur Blvd, Oakland, CA | Driller | Jesse Pattison | | | _ | | |
| Project Manager: Lynelle | Onishi | Type o | f Drilling Rig: 1 | Marl M | 2.5 D |)P | <u> </u> | |
| RG: John McCain | | Drilling | g Method: Direc | t Push | | | | |
| Geologist: John McCain | | Sampl | ing Method: Co | intinuo | us M | acto-Con | e with | acetate sleeve. |
| JOD NUMDER: 38487349.0/ | | Date(s |) Drilled: 07/14 | /05 | | | | |
| Conversion to a Desetter De | BORING IN | FORMA | TION | | <u> </u> | 1 11 1 100 | | |
| Air Katte an Head Auges | Deaths 5 0 feeth and 1 have | Boring | Location: South | west o | d used | d oil UST | - | |
| Air Knite or Hand Auger | Depth: 5.0 test bgs/Hand Auger | Boring | Diameter: 2-ind | | | | | |
| Coordinates: X | T | Boring | Type: Explorate | ory | | | | |
| Depth (ft bgs) Symbol | Lithologic Description | n | | nscs | PID (ppm) | Sample ID | Recovery | Comments |
| | aserock: AC cover (4") with baserock (2") | beneath | | FILL | <u> </u> | | | |
| SiLTY sand, to 0.2 | CLAYEY SAND: FILL, very dark gray bro 7.5% silt, 7.5% clay, 5% gravel, fine sand 5" diameter, loose, moist to wet, petroleum | own (Gley ' is, trace an n odor | l 3/10Y), 80% gular gravels | | 000 | 60.7 | | |
| SILTY 15% s plastic | Y SANDY CLAY: FILL, dark gray brown (G sand, 10% silt, 5% gravels, fine sands, sof city, petroleum odor | /: FILL, dark gray brown (Gley 1 3/10Y), 70% clay, 5% gravels, fine sands, soft, moist to wet, med. odor | | | | (2-2.5') | - | |
| 6 | | | | | 429 | SB-7 (5-5.5°) | | |
| SILTY | SANDY CLAY: light olive brown (2.5Y 5/ and, fine sands, stiff, molst, med. plasticity | 3), 85% cla , no petrol | ay, 10% siit, sum odor | CL | | | | |
| 10 @ 10' same odor | - Sandy Silty Clay continues, 60% clay, 1 color as above, stiff, moist, med. plasticity | y Silty Clay continues, 60% clay, 10% slit, 10% sand, s above, stiff, moist, med. plasticity, no petroleum | | | 7.5 | 88-7 (9.5-10') | | |
| - 14 SANE sand, | Y CLAYEY SILT: clive brown (2.5Y 4/3), i fine sands, stiff, moist, low plasticity, no p | 80% silt, 1 etroleum c | 0% clay, 10% dor | ML | 1.6 | SB-7 (14.5-15) | | |
| - 18 | | | | | Û.E. | SB-7 (18.5-17) SB-7 | | |
| - 20 @ 20' 80% s patrol | - Sandy Clayey Silt continues, light olive t silt, 10% clay, 10% sand, very stiff, moist, l eum odor | brown (2.5 low plastic | Y 5/4), ty, no | | Ç.U | (19.5-20") | | |
| | | | | u (i | - | i Involució | | 0D 7 |

| UR | | LOG OF BORING | E | Bore | hole l | D:S | B- 7 |
|----------------|--------|--|----------------------|-------------------|--|----------|-------------|
| Depth (ft bgs) | Symbol | Lithologic Description | uscs | (mqq) CII | Sample I.D. | Recovery | Comments |
| 26 | | 24.5' - Sandy Clayey Sitt continues, light olive brown (2.5Y 5/4), 80% sitt, 10% clay, 10% sand, very stiff, molat, low plasticity, no petroleum odor SILTY SAND: brown (2.5Y 4/3), 90% sand, 10% silt, fine sands, loosa, moist to wet, no petroleum odor SILTY SANDY CLAY: olive brown (2.5Y 4/3), 80% clay, 10% silt, 10% sand, fine sands, med. stiff, moist, med. plasticity, no petroleum odor SILTY SAND: brown (2.5Y 5/4), 90% sand, 10% silt, fine sands, loose, wet, no petroleum odor SILTY SAND: brown (2.5Y 5/4), 90% sand, 10% silt, fine sands, loose, wet, no petroleum odor SILTY SANDY CLAY: brown (2.5Y 4/3), 80% clay, 10% silt, 10% sand, fine sands, med. stiff, moist, med. plasticity, no petroleum odor SILTY SANDY CLAY: brown (2.5Y 4/3), 80% clay, 10% silt, 10% sand, fine sands, med. stiff, moist, med. plasticity, no petroleum odor | SM CL SM CL | 4.2 0.1 0.1 | SB-7 (25.5-26") SB-7 (28.5-29") SB-7 (30.5-31") Borehole grouted to grade with | | ¥ |

BP/Atlantic Richfield Company Page 2 of 2

Borehole ID: SB-7

| | | | | 1.04 | 2 ~ | | | | | |
|--|--------------------------|---|-----------------------|-------------------|------------|-------------|--------------------|----------|-----------------|--|
| | | 1333 Broadway, Suite 80 | 00 | LU(| <u>J U</u> | | SURI | NG | | |
| | | Oakland, California 9461 | 2 | Total Dopth | 1: 31 | 5-8 44 % | | | | |
| | | | | | . 20 | πD | gs | | | |
| Project: BP f | PROJECT INFORMATION DRIL | | | DRILL | | | | | | |
| Site L acction: 100 Mash thus Divel Onlined CA | | | Driller | g Company: Gri | gg LJ | niiinį | , & Testir | lg | | |
| Project Mana | der:) | wielle Onishi | | Prilling Bla: M | far! M | 17 S T | | | | |
| RG: John McC | ain | | Driliin | a Method: Direct | Push | | ···· | | | |
| Geologist: Jo | hn Mo | Cain | Sampl | ing Method: Co | ntinuç | us M | acro-Core | e with a | acetate sleeve. | |
| Job Number: | 3848 | 7349.0A034 | Date(s |) Drilled: 07/13/ | 05 | | | | ····· | |
| | | BORING INF | ORMA | TION | | | | | | |
| Groundwater | Dept | h: 7 ft bgs | Boring | Location: East o | fMŴ | -1 | | | | |
| Air Knife or I | land . | Auger Depth: 5.0 feet bgs/Hand Auger | Boring | Diameter: 2-incl | h | | | | | |
| Coordinates: | X | Y | Boring | Type: Explorator | ry | | | | | |
| (S | | | | | | | | | | |
| ft bg | 5 | | | | ģ | Ed | 9 0 | Ую, | | |
| ب ج | dmy | Lithologic Description | l | | Sc | | du | S | Comments | |
| Dep | ů, | | | | _ | 료 | Sa | Ľ. | | |
| <u> </u> | <u> </u> | | | | | L |] | 1 | | |
| | | AC/Baserock: AC cover (4") with baserock (2") b SILTY GRAVELY SAND: EII office (5YR 4/4) | eneath 80% sand | 5% ellt 5% | FILL | | | | | |
| | | clay, 10% gravel, fine to coarse sands, trace sub | -angular (| gravels to | | | | | | |
| |] | | | | | | | | | |
| ~ | | | | 1 | | | | | | |
| – 4 | | | | | | | | | | |
| | | | | | | 0.4 | SB-8 | | | |
| - 6 | | 17 - Silly Cand poor (2" thick) with some in St | N/ acred | 1.00/ -514 | | | (5-5.5') | | | |
| | ł | 10% gravels, reddish brown (2.5YR 5/4), angular | r gravels t | o 0.25*, | | İ | | | V | |
| - 8 | | SILTY CLAYEY SAND: brown (2.5YR 4/4), 75% | sand, 10 ⁴ | % silt, 10% | SM | 0,0 | (7-7.5°) | | | |
| | 12 | clay, 5% gravels, fine sands, dense, moist, no pe | stroleum d | odor | | | | Kanas | | |
| Ē | -22 | | | | | 24.1 | SB-8 | | | |
| - 10 E | -22 | | | | | | (9.5-10') | | | |
| | <u> </u> | @ 11' - Silty Sand seam (3" thick) with gravels, 8 | 10% sand, | 10% silt. | | 0,0 | SS-8 | | | |
| - 12 | -44 | wet, no petroleum odor | io 0.25", | idose, | | | 1 (11-11.0) | | | |
| | -44 | | | | | | | | | |
| <u>–</u> 14 | <u> </u> | | | | | | | | | |
| | <u>~</u> | | | | | 0.1 | SB-8 (14.5-15) | | | |
| E 16 | <u>~</u> | | | | | | | | | |
| | <u></u> | | | | | ŀ | | | | |
| 40 | <u>~</u> | @ 17.5' - Silty Clayey Sand seam (6" thick). 80% | sand. 10 | % siit, | | 0.0 | SB-8 | | | |
| | | 10% cley, reddish brown (2.5YR 5/4), loose, wet, odor | no petrol | eum | | | (17.5-18') | | | |
| - | | | | | | | 60.6 | | | |
| 20 | ` | | | | | U,U | (19.5-20) | | | |
| | <u>></u> | | | | | 0,0 | SB-8 (20.5-21') | | | |
| - 22 | <u> </u> | | | | | | | | | |
| | -22 | | | | | | | | | |
| E. 24 | <u></u> | | | | | | | | | |
| | and | p Dichfield Company: | al 2 | | l | F | orehol | » (D · | SP-8 | |
| DF/All | שווע | | | | | ندا | | | | |

| UR | S LOG OF BORING | F | Bore | hole l | D:S | B-8 |
|--------------------------|---|------|------------------|--|----------|----------|
| Depth (ft bgs) Sumbol | Lithologic Description | USCS | (mqq) Qi4 | Sample I.D. | Recovery | Comments |
| 26 | SILTY CLAY: reddish brown (5YR 5/3), 85% clay, 10% silt, 5% sand, trace fine sands, stiff, molst, med. plasticity, no petroleum odor | CL | | Borehole grouted to grade with neat Portland coment | | |
| 30 | Depth discrete groundwater samples were attempted within a boring 1 foot laterally from this location and were not successful. | | - - - - | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| BP/Atlan | tic Richfield Company Page 2 of 2 | | B | orehole | ID : | SB-8 |

ARCADIS

Appendix B

Concentration versus Time Graphs for TPHg, Benzene and MTBE



2. Non-detected analytical results are graphed at the laboratory reporting limit.

3. For additional information about data for a given sampling event (such as no data plotted), refer to the historical analytical table.

3/12/2012



2. Non-detected analytical results are graphed at the laboratory reporting limit.

3. For additional information about data for a given sampling event (such as no data plotted), refer to the historical analytical table.

3/12/2012



2. Non-detected analytical results are graphed at the laboratory reporting limit.

3. For additional information about data for a given sampling event (such as no data plotted), refer to the historical analytical table.

3/12/2012



2. Non-detected analytical results are graphed at the laboratory reporting limit.

3. For additional information about data for a given sampling event (such as no data plotted), refer to the historical analytical table.

3/12/2012

ARCADIS

Appendix C

Historical Underground Utility Figure



SOURCE: Pacific Gas and Electric Company, Pacific Bell, East Bay Municipal Utility District. City of Oakland. Alameda County Assessor's Office

ARCADIS

Appendix D

Alameda County Environmental Health Case Closure Summary

CASE CLOSURE SUMMARY LEAKING UNDERGROUND FUEL STORAGE TANK – LOCAL OVERSIGHT PROGRAM

I. AGENCY INFORMATION

Date: March 12, 2012

| Agency Name: Alameda County Environmental Health | Address: 1131 Harbor Bay Parkway |
|--|---------------------------------------|
| City/State/Zip: Alameda, CA 94502-6577 | Phone: 510.668.4411 |
| Responsible Staff Person: Paresh Khatri | Title: Groundwater Resources Engineer |

II. CASE INFORMATION

| Site Facility Name: Former BP Service Station #11102 | | | | | |
|---|--|--|----------------|--|--|
| Site Facility Address: 100 MacArthur Boulevard, Oakland, CA | | | | | |
| RB Case No.: 01-0985 | Local Case No.: RO0000456 LOP Case No.: RO0000456 | | | | |
| URF Filing Date: | URF Filing Date: Global ID No.: T0600100908 APN: 010-0812-008-01 | | | | |
| Responsible Parties Addresses | | | Phone Numbers | | |
| Atlantic Richfield Company P.O. Box 1257, San Ramon, CA 94583 (925) 275-3 | | | (925) 275-3801 | | |

| Tank I.D. No | Size in Gallons | Contents | Closed In Place/Removed? | Date |
|--------------|-----------------|------------------------------|-----------------------------|--------------|
| Unknown | 12,000 | Regular Unleaded Gasoline | In Place | 1990-Present |
| Unknown | 10,000 | Unleaded Plus Gasoline | In Place | 1990-Present |
| Unknown | 6,000 | Super Unleaded Gasoline | In Place | 1990-Present |
| Unknown | 1,000 | Waste Oil | In Place | 1988-Present |
| Unknown | 6,000 | Gasoline | Removed | 1990 |
| Unknown | 6,000 | Gasoline | Removed | 1990 |
| Unknown | 8,000 | Gasoline | Removed | 1990 |
| Unknown | 550 | Waste Oil | Removed | 1988 |

Conflicting information exists regarding the previous USTs at the site. According to Mobile, one 6,000-gallon, one 8,000-gallon and one 10,000-gallon USTs were installed at the site in 1983 (Mobile, 1987). Additionally, Mobile also documents that one 12,000-gallon, one 10,000-gallon and one 6,000-gallon USTs were installed in 1982 (Mobile, 1986 and undated). The waste oil tank removed in 1988 is documented as being a 280-gallon capacity by Mobile (Mobile, 1989). The same waste oil UST is described as being 550-gallong capacity by KEI (KEI, 1988). The information provided above is based up on the records from the Alameda County Environmental Health Department.

III. RELEASE AND SITE CHARACTERIZATION INFORMATION

| Cause and Type of Release: Exact release source is unknown; upon excavation in 1988 the Waste Oil UST was observed having a hole in the tank and petroleum product was observed "dripping" from the west sidewall of the excavation (ACDEH, 1988). | | | | | | |
|--|------|---|---|--|--|--|
| Site characterization complete? Yes | Date | Approved By Oversight | Agency: TBD | | | |
| Monitoring wells installed? Yes | | Number: 4 | Proper screened interval? yes | | | |
| Highest GW Depth Below Ground Surface: 17 feet (below top of casing) | .97 | Lowest Depth: 8.02 feet (below top of casing) | Flow Direction: predominately west/southwest | | | |
| Most Sensitive Current Use: Municipal wells | | | | | | |

| Summary of Production Wells in Vicinity: Production wells are not present in site vicinity (1/4 mile) | | | | | | |
|---|----------------------------------|--|--|--|--|--|
| Are drinking water wells affected? No | Aquifer Name: NA | | | | | |
| Is surface water affected? No | Nearest SW Name: Glen Echo Creek | | | | | |
| Off-Site Beneficial Use Impacts (Addresses/Locations): None | | | | | | |
| Reports on file? Yes with ACEH Where are reports filed? ACEH & SFBRWQCB | | | | | | |

| TREATMENT AND DISPOSAL OF AFFECTED MATERIAL | | | | | | |
|---|--|--|--|--|--|--|
| Material | Amount (Include Units) | Date | | | | |
| Tank | One 8,000-gallon UST Two 6,000-gallon UST One 550-gallon UST | 8,000-gallon Gasoline UST and 6,000- Gasolie USTs removed in 1990; 550-gallon Waste Oil USTs removed in 1988. Limited documentation from UST removals exist | Gasoline USTs-1990 Waste Oil UST-1988 | | | |
| Piping | Unknown | | | | | |
| Free Product | None | | | | | |
| Soil | 15 Cubic Yards | Soil removed in September 1988 during Waste Oil UST removal | September 19, 1988 | | | |
| Groundwater | Unknown | | | | | |

MAXIMUM DOCUMENTED CONTAMINANT CONCENTRATIONS BEFORE AND AFTER CLEANUP (Please see appendices for additional information on contaminant locations and concentrations)

| Contominant | Soil (j | ppm) ¹ | Water (ppb) | | | | | |
|----------------------|--|--|---|---|--|--|--|--|
| Contaminant | Before | After ² | Before | After ² | | | | |
| TPHg | 1,300 mg/kg (SB-7@ 2.0', 7/4/2005) | <0.3 mg/kg (MW-4@ 6.5', 10/6/2010) | 45,000 • g/L MW-2, 2/19/2003 | 1,900 ● g/L MW-3, 8/16/2010 | | | | |
| TPHd | 2,100 mg/kg (TD-1@ 0.5', 11/22/1994) | 2,100 mg/kg (TD-1@ 0.5', 11/22/1994) | 2,900 ∙ g/L MW-1, 6/18/1998 | 160 ∙ g/L MW-1, 7/10/2007 | | | | |
| Total Oil and Grease | 24 mg/kg (WO @ 9', 9/19/1988) | <1.0 mg/kg (THP-2 @ 10', 11/22/1994) | 14,000 ∙ g/L MW-1, 3/1/1991 | <5.0 ● g/L MW-1, 6/18/1998 | | | | |
| Benzene | 0.08 mg/kg (MW-2@ 10', 10/25/1989) | <0.005 mg/kg (MW-4@ 6.5', 10/6/2010) | 440 ∙ g/L MW-1, 5/19/1992 | 1.6 ∙ g/L MW-1, 5/7/2009 | | | | |
| Toluene | 0.006 mg/kg (MW-3@ 5', 10/26/1989) | <0.005 mg/kg (MW-4@ 6.5', 10/6/2010) | 110 ∙ g/L MW-1, 2/19/2003 | 1.0 ∙ g/L MW-1, 4/21/2006 | | | | |
| Ethylbenzene | 3.0 mg/kg (SB-7@ 2.0', 7/14/2005) | <0.005 mg/kg (MW-4@ 6.5', 10/6/2010) | 250 ∙ g/L MW-1, 5/19/1992 | 0.83 ∙ g/L MW-1, 1/22/2008 | | | | |
| Xylenes | 3.9 mg/kg (SB-7@ 5', 7/14/2005) | <0.01 mg/kg (MW-4@ 6.5', 10/6/2010) | 1,100 ∙ g/L MW-1, 3/9/1999 | 0.89 • g/L MW-1, 9/3/2009 | | | | |
| МТВЕ | 0.37 mg/kg (SB-4@ 9.5', 7/14/2005) | <0.005 mg/kg (MW-4@ 6.5', 10/6/2010) | 49,000 • g/L ³ MW-1, 3/9/1999 | 2,600 • g/L ⁴ MW-3, 8/22/2011 | | | | |
| Lead | Not analyzed | Not analyzed | Not analyzed | Not analyzed | | | | |

¹ Soil results are from samples collected at depths of 10 feet bgs or shallower. ² After results are represented by the maximum concentration.

³Other fuel oxygenates (groundwater [μg/L] before cleanup): TBA 11,000 μg/L (MW-2), DIPE <0.50 μg/L, ETBE <0.5 μg/L, TAME 420 µg/L (MW-2), 1.2-DCA 2.8 µg/L (MW-2), ethanol 120 µg/L (MW-1)

⁴ Other fuel oxygenates (groundwater [µg/L] after cleanup): TBA 2,600 µg/L (MW-2), DIPE <0.50 µg/L, ETBE <0.50 µg/L, TAME 3.9 µg/L (MW-2), 1.2-DCA 2.8 µg/L (MW-2), ethanol 120 µg/L (MW-1)

IV. CLOSURE

Does completed corrective action protect existing beneficial uses per the Regional Board Basin Plan? Yes

Does completed corrective action protect potential beneficial uses per the Regional Board Basin Plan? Yes

Does corrective action protect public health for current land use? It does not appear that the release presents a significant risk to human health based upon current land use and conditions.

Site Management Requirements: City of Oakland Building Department will be notified if excavation or development of the site be proposed that impacted soil or groundwater may be encountered, Alameda County Environmental Health must be notified as required by Government Code Section 65850.2.2.

Should corrective action be reviewed if land use changes? It would depend on when the land use changes (site could be fully naturally attenuated at that time) and what the proposed land use change was.

| Was a deed restriction or deed notification filed | Date Recorded: | |
|--|----------------------------|-----------------------|
| Monitoring Wells Decommissioned: Upon Case Closure Approval | Number Decommissioned: All | Number Retained: None |
| List Enforcement Actions Taken: NA | | |
| List Enforcement Actions Rescinded: NA | | |

Considerations and/or Variances:

Low levels of residual soil and groundwater contamination remain onsite (refer Tables 1 and 2 below); however, these concentrations do not pose a significant risk because the entire site is capped and the groundwater plume is contained on the site.

Conclusion:

ARCADIS requests approval for case closure and no further action at this site based on the following:

- Free product has never been observed in any of the monitoring wells or excavated areas on-site.
- Sources of petroleum hydrocarbons at the site have been adequately addressed. The absence of high
 concentrations of site COCs observed in soil samples suggests that secondary sources (residual hydrocarbons
 in soil) were identified and removed.
- Current groundwater concentrations are orders of magnitude below historical maximums, are continuing to decrease with time, or are not detected above the detection limit, indicating that natural attenuation processes are occurring at the Site.
- The Site has been adequately characterized.
- The low-level plume is decreasing in size and is generally confined to the site boundaries.
- Concentrations of BTEX in groundwater are all below the reporting limits.
- Concentrations of TPHg in groundwater are all below reporting limits.
- All the wells that contain MTBE concentrations in groundwater indicate a decreasing trend.
- All the wells that contain TBA concentrations in groundwater indicate a decreasing trend.
- Current site conditions suggest that soil concentrations of TPHg are limited to the western portion of the site in the vicinity of SB-7 (**Figure 7**).
- Current site conditions suggest that soil concentrations of MTBE are limited to the western portion of the site in the vicinity of SB-7 (**Figure 7**).
- No sensitive receptors are likely to be impacted, including surface-water bodies, municipal wells and drinking water sources.
- The site presents no current or potential risk to human health or the environment.

VI. LOCAL AGENCY REPRESENTATIVE DATA

| Prepared by: Paresh Khatri | Title: Hazardous Materials Specialist | | |
|----------------------------|---------------------------------------|--|--|
| Signature: | Date: | | |
| Approved by: | Title: | | |
| Signature: | Date: | | |

This closure approval is based upon the available information and with the provision that the information provided to this agency was accurate and representative of site conditions.

VII. REGIONAL BOARD NOTIFICATION

| Regional Board Staff Name: Cherie McCaulou | Title: Engineering Geologist | | |
|--|------------------------------|--|--|
| RB Response: | Date Submitted to RB: | | |
| Signature: | Date: | | |

VIII. MONITORING WELL DECOMMISSIONING

| Date Requested by ACEH: TBD | Date of Well Decommissioning Report: TBD | | | | | |
|--|---|--|--|--|--|--|
| All Monitoring Wells Decommissioned: TBD | TBD Number Decommissioned: TBD Number Retained: TBD | | | | | |
| Reason Wells Retained: TBD | | | | | | |
| Additional requirements for submittal of groundwater data from retained wells: | | | | | | |
| ACEH Concurrence - Signature: | Date: | | | | | |

Attachments:

- 1. Tables 1 and 2 (comparison of maximum residual contamination to applicable ESLs or approved cleanup goals).
- 2. Site figures provided in Case Closure Summary Report.
- 3. Analytical data tables for soil, groundwater, depth to groundwater, etc. are provided in Case Closure Summary Report.
- 4. Boring logs/monitoring well construction details are provided in Appendix A of the Case Closure Summary Report.

This document and the related CASE CLOSURE LETTER & REMEDIAL ACTION COMPLETION CERTIFICATE will be retained by the lead agency as part of the official site file.

Environmental Impacts in Groundwater BP 11102 100 MacArthur Blvd, Oakland, California

Table 1. Comparison of Most Current Maximum Residual Soil Concentrations at the Site to Relevant Cleanup Standards (mg/kg)

| | TPHg (mg/kg) | Benzene (mg/kg) | Toluene (mg/kg) | Ethylbenzene (mg/kg) | Xylenes (mg/kg) | MTBE (mg/kg) | TPH-DRO (mg/kg) |
|---|--|--|--|--|---|--|--|
| Current Maximum Residual Soil Concentrations at Site ^{1,3} | <0.3 mg/kg (MW-4@ 6.5', 10/6/2010) | <0.005 mg/kg (MW-4@ 6.5', 10/6/2010) | <0.005 mg/kg (MW-4@ 6.5', 10/6/2010) | <0.005 mg/kg (MW-4@ 6.5', 10/6/2010) | <0.01 mg/kg (MW-4@ 6.5', 10/6/2010) | <0.005 mg/kg (MW-4@ 6.5', 10/6/2010) | 2,100 mg/kg (TD-1@ 0.5', 11/22/1994) |
| RWQCB, Region 2 ESLs ¹ | 83 ² | 0.044 ² | 2.9 ² | 3.3 ² | 2.3 ² | 0.023 ² | 83 ² |

Notes:

¹ Regional Water Quality Control Board, San Francisco Bay Region (RWQCB) Environmental Screening Levels (ESLs) Screening for Environmental Concerns at Sites with Contaminated Soil and Groundwater, Water Board, May 2008.

² ESL value for Shallow Soils (< 3 meters bgs) where groundwater is a current or potential source of drinking water-commercial/industrial land use (Table A).

³ Depth to water ranges between 8.02 feet and 17.97 feet below top of casing; therefore, all soil results are from sample depths of 10 feet or shallower.

Environmental Impacts in Groundwater BP 11102 100 MacArthur Blvd, Oakland, California

Table 2. Comparison of Most Current Maximum Residual Groundwater Concentrations at the Site to Relevant Cleanup Standards (µg/L)

| | TPH-GRO (µg/L) | Benzene (µg/L) | Toluene (μg/L) | Ethylbenzene (µg/L) | Xylenes (µg/L) | MTBE (µg/L) | TPH-DRO (µg/L) |
|---|---|--|--|--|---|-------------------------------|-------------------|
| Maximum Residual Groundwater Concentrations at Site | <1,000 (all active monitoring wells) | <0.50 (all active monitoring wells) | <0.50 (all active monitoring wells) | <0.50 (all active monitoring wells) | <1.0 (all active monitoring wells) | 1,700 (MW-3, 2/20/2012) | |
| RWQCB, Region 2 ESLs ¹ | 100 ² | 1.0 ² | 40 ² | 30 ² | 20 ² | 5.0 ² | 100 ² |

Notes:

¹ Regional Water Quality Control Board, San Francisco Bay Region (Water Board) Environmental Screening Levels (ESLs) Screening for *Environmental Concerns at Sites with Contaminated Soil and Groundwater*, Water Board, May 2008.

² ESL value for Shallow Soils (< 3 meters bgs) where groundwater is a current or potential source of drinking water-commercial/industrial land use (Table A).